IN THE DISTRICT COURT OF THE VIRGIN ISLANDS BANKRUPTCY DIVISION ST. CROIX, VIRGIN ISLANDS

In re:)	
m ic.)	Chapter 11
HOVENSA L.L.C.,)	•
)	Case No. 1:15-bk-10003-MFW
	Debtor.)	
)	Re: Docket Nos. 15 and 220

ORDER (A)(I) APPROVING THE SALE OF THE DEBTOR'S ASSETS, FREE AND CLEAR OF ALL LIENS, CLAIMS, ENCUMBRANCES, AND INTERESTS; AND (II) AUTHORIZING THE ASSUMPTION AND ASSIGNMENT OF CERTAIN EXECUTORY CONTRACTS AND UNEXPIRED LEASES; AND (B) GRANTING RELATED RELIEF

Upon the motion, dated September 15, 2015 (the "Motion"), of HOVENSA L.L.C., as debtor and debtor-in-possession (the "Debtor") in the above-captioned chapter 11 case (the "Chapter 11 Case"), pursuant to sections 105, 363, 365, 503, and 507 of title 11 of the United States Code (the "Bankruptcy Code"), Rules 2002, 6004, 6006, and 9014 of the Federal Rules of Bankruptcy Procedure (the "Bankruptcy Rules"), and Local Rules 6004-1 and 6006-1, for (i) entry of an order (the "Bidding Procedures Order"): (a) approving the proposed auction and bidding procedures (the "Bidding Procedures"), including approval of the Break-Up Fee and the Expense Reimbursement (as such terms are defined in the Motion); (b) establishing procedures for the assumption and assignment of executory contracts and unexpired leases, including notice of proposed cure amounts (the "Assumption and Assignment Procedures"); (c) approving the form and manner of notice of all procedures, protections, schedules, and agreements; and (d) scheduling an auction (the "Auction") and a hearing (the "Sale Hearing") to approve the sale transaction related to the Debtor's crude oil and product storage and terminalling business (the "Sale"); and (ii) entry of an order (the "Sale Order"): (a) approving the Sale of the Debtor's assets, free and clear of all Interests (as defined in paragraph V below);

Case 1:15-bk-10003-MFW Doc 394 Filed 12/01/15 Entered 12/01/15 12:35:34 Desc Main Document Page 2 of 35

(b) authorizing the assumption and assignment of certain executory contracts and unexpired leases; and (c) granting certain related relief; after holding a hearing on October 8, 2015 (the "Bidding Procedures Hearing"), this Court having entered the Bidding Procedures Order on October 9, 2015 (Docket No. 220); and the Auction having been commenced on November 10, 2015 and thereafter closed on November 19, 2015; and Limetree Bay Holdings, LLC or its permitted designee (the "Buyer") having been selected as the Successful Bidder; and upon the Buyer, the Debtor, and Hess Oil Virgin Islands Corp. ("HOVIC") having entered into that certain Amended and Restated Asset Purchase Agreement in substantially the form attached hereto as Exhibit A, and as it may be amended, modified, or supplemented in accordance with the terms hereof and thereof (the "Asset Purchase Agreement"); and this Court having conducted the Sale Hearing to consider entry of this Sale Order on November 17, 19, and 30, 2015; and all parties in interest having been heard, or having had the opportunity to be heard, regarding the Motion, the Asset Purchase Agreement, and this Sale Order; and this Court having reviewed and considered the Motion and all objections thereto, and the arguments of counsel made, and the evidence adduced, at the Bidding Procedures Hearing and the Sale Hearing; and upon the entire record of the Bidding Procedures Hearing and the Sale Hearing, and after due deliberation thereon, and good cause appearing therefor:

Capitalized terms used, but not otherwise defined, herein shall have the meaning ascribed to such terms in the Asset Purchase Agreement or, if not defined therein, in the Motion.

Jurisdiction, Final Order and Statutory Predicates

A. This Court has jurisdiction to hear and determine the Motion under 28 U.S.C. §§ 157 and 1334. This is a core proceeding under 28 U.S.C. § 157(b). Venue of this Chapter 11 Case and the Motion in this district is proper under 28 U.S.C. §§ 1408 and 1409.

B. This Sale Order constitutes a final order within the meaning of 28 U.S.C. § 158(a). Notwithstanding Bankruptcy Rules 6004(h) and 6006(d), and to any extent necessary under Bankruptcy Rule 9014 and Rule 54(b) of the Federal Rules of Civil Procedure, as made applicable by Bankruptcy Rule 7054, this Court expressly finds that there is no just reason for delay in the implementation of this Sale Order, and expressly directs entry of judgment as set forth herein.

C. The statutory predicates for the relief sought in the Motion are sections 105(a), 363, 365, 503, and 507 of the Bankruptcy Code, Bankruptcy Rules 2002, 6004, 6006 and 9014, and the applicable Local Rules of Bankruptcy Practice and Procedure of the United States Bankruptcy Court for the District of the Virgin Islands (the "Local Rules").

Notice of the Sale, Auction and the Cure Payments

D. Actual written notice of the Sale Hearing, the Auction, the Motion, the Sale, and the assumption, assignment and/or transfer of the Assigned Contracts (as defined in paragraph F below), and a reasonable opportunity to object or be heard with respect thereto and to the entry of this Sale Order has been afforded to all known interested Persons and entities entitled to receive such notice, including, but not limited to, the following parties: (i) counsel to the

² The findings and conclusions set forth herein constitute this Court's findings of fact and conclusions of law pursuant to Bankruptcy Rule 7052, made applicable to this proceeding pursuant to Bankruptcy Rule 9014. To the extent that any of the following findings of fact constitute conclusions of law, they are adopted as such. To the extent any of the following conclusions of law constitute findings of fact, they are adopted as such.

Case 1:15-bk-10003-MFW Doc 394 Filed 12/01/15 Entered 12/01/15 12:35:34 Des Main Document Page 4 of 35

Committee; (ii) counsel to the Buyer; (iii) the Office of the United States Trustee for the District of the U.S. Virgin Islands; (iv) all entities (or counsel therefore) known to have asserted any lien, charge, claim or encumbrance on the Purchased Assets; (v) all federal, state and local regulatory or taxing authorities which are reasonably ascertainable by the Debtor to have a known interest in the Purchased Assets; (vi) all known non-debtor counterparties to any unexpired leases or executory contracts that could potentially be assumed and assigned to the successful bidder; (vii) those parties who expressed a bona fide interest in acquiring the Assets in the six months preceding the Petition Date; (viii) the Internal Revenue Service; (ix) those parties who have made the appropriate filings requesting notice of all pleadings filed in the chapter 11 case; (x) the United States Attorney General's Office for the District of the Virgin Islands; (xi) Hess Oil Virgin Islands Corporation; (xii) PDVSA, V.I., Inc.; and (xiii) such other entities as may be required by applicable Bankruptcy Rules or applicable Local Rules or as may be reasonably requested by the Buyer.

- E. The Debtor published the Auction and Sale Notice in each of *The Financial Times*, *The Wall Street Journal*, and *The Virgin Islands Daily News* on or about October 14, 2015.
- F. In accordance with the provisions of the Bidding Procedures Order, the Debtor has served notice (the "Assumption and Assignment Notice") upon all of the non-debtor counterparties (the "Contract Counterparties") to the unexpired leases and executory contracts being assumed and assigned to the Buyer (the "Assigned Contracts") setting forth: (i) the contract(s) and/or lease(s) that may be assumed by the Debtor and assigned to the Buyer; (ii) the name and address of the Contract Counterparty thereto; (iii) notice of the proposed effective date of the assignment (subject to the right of the Debtor and/or the Buyer to withdraw such request for assumption and assignment of the Assigned Contract(s) prior to the Closing); (iv) the amount, if any, determined by the Debtor to be necessary to be paid to cure and compensate for

Case 1:15-bk-10003-MFW Doc 394 Filed 12/01/15 Entered 12/01/15 12:35:34 Desc Main Document Page 5 of 35

any existing default in accordance with sections 365(b) and 365(f)(2) of the Bankruptcy Code

(the "Cure Amount"); and (v) the deadlines by which any such Contract Counterparty must file

an objection to the proposed assumption and assignment of any Assigned Contract.

G. The service of such Assumption and Assignment Notice (i) was good, sufficient and

appropriate under the circumstances of this Chapter 11 Case; (ii) provided such counterparties

with a full and fair opportunity to object to such assumption, assignment, or transfer and to the

proposed Cure Amount set forth in the Assumption and Assignment Notice; and (iii) was in

compliance with the Bidding Procedures Order and applicable provisions of the Bankruptcy

Rules and Local Rules. Accordingly, no other or further notice need be given in connection with

such assumption, assignment, or transfer or with respect to the amount of Cure Payments.

H. As evidenced by the affidavits of service previously filed with this Court and as

approved under the Bidding Procedures Order: (i) due, proper, timely, adequate and sufficient

notice of the Motion, the Auction, the Sale Hearing, the assumption and assignment of the

Assigned Contracts, the entry of this Sale Order, and the Sale has been provided to all parties-in-

interest; (ii) such notice was, and is, good, sufficient and appropriate under the circumstances of

this Chapter 11 Case, provided a fair and reasonable opportunity for parties-in-interest to object,

and to be heard, with respect thereto, and was provided in accordance with sections 102(1), 363

and 365 of the Bankruptcy Code, Bankruptcy Rules 2002, 6004, 6006, 9006, 9007 and 9014, and

the applicable Local Rules; and (iii) no other or further notice of with respect to such matters is

necessary or shall be required.

Business Judgment

I. The Debtor has demonstrated good, sufficient and sound business purposes and

justifications for, and compelling circumstances to promptly consummate, the Sale and other

transactions contemplated by the Asset Purchase Agreement and the Transaction Documents,

5

Case 1:15-bk-10003-MFW Doc 394 Filed 12/01/15 Entered 12/01/15 12:35:34 Desc Main Document Page 6 of 35

including, without limitation, the assumption, assignment, and/or transfer of the Assigned Contracts (collectively, the "<u>Transactions</u>") pursuant to sections 363 and 365 of the Bankruptcy Code, prior to and outside of a plan of reorganization, and such action is an appropriate exercise of the Debtor's business judgment and in the best interests of the Debtor, its Estate, and its creditors.

Good Faith of the Buyer; No Collusion

J. The Buyer is not an insider (as that term is defined in section 101(31) of the Bankruptcy Code) of the Debtor.

K. The Buyer is purchasing the Purchased Assets in good faith, and is a good faith purchaser, within the meaning of section 363(m) of the Bankruptcy Code. The Buyer is therefore entitled to, and granted pursuant to paragraph 24 below, the full rights, benefits, privileges, and protections of section 363(m) of the Bankruptcy Code, and the Buyer has otherwise proceeded in good faith in all respects in connection with the Transactions in that, *inter alia*: (i) the Buyer recognized that the Debtor was free to deal with any other party interested in acquiring the Purchased Assets; (ii) the Buyer complied with the provisions in the Bidding Procedures Order; (iii) the Buyer agreed to subject its bid to the competitive bidding procedures set forth in the Bidding Procedures Order; (iv) all payments to be made by the Buyer and other agreements or arrangements entered into by the Buyer in connection with the Sale have been disclosed; (v) the Buyer has not violated section 363(n) of the Bankruptcy Code by any action or inaction; (vi) no common identity of directors or controlling stockholders exists between the Buyer, on the one hand, and the Debtor, on the other hand; and (vii) the negotiation and execution of the Asset Purchase Agreement and Transaction Documents were at arms' length and in good faith.

Case 1:15-bk-10003-MFW Doc 394 Filed 12/01/15 Entered 12/01/15 12:35:34 Desc Main Document Page 7 of 35

L. None of the Debtor, the Buyer, or any of their respective Representatives, has engaged in any conduct that would cause or permit the Asset Purchase Agreement or any of the Transaction Documents, or the consummation of the Transaction, to be avoidable or avoided, or for costs or damages to be imposed, under section 363(n) of the Bankruptcy Code, or has acted in bad faith or in any improper or collusive manner with any Person in connection therewith.

Highest and Best Offer

M. In accordance with the Bidding Procedures Order, the Asset Purchase Agreement was deemed a Qualified Bid (as defined in the Bidding Procedures Order) and the Buyer was a Qualified Bidder (as defined in the Bidding Procedures Order) eligible to participate at the Auction.

N. The Debtor conducted a marketing and auction process in accordance with, and has otherwise complied in all respects with, the Bidding Procedures Order. The marketing and auction process set forth in the Bidding Procedures Order afforded a full, fair and reasonable opportunity for any Person or entity to make a higher or otherwise better offer to purchase the Purchased Assets. The Auction was duly noticed and conducted in a non-collusive, fair and good faith manner and a reasonable opportunity has been given to any interested party to make a higher or otherwise better offer for the Purchased Assets.

O. As qualified by the Court's findings made on the record at the Sale Hearing, the Asset Purchase Agreement constitutes the highest or otherwise best offer for the Purchased Assets, and will provide a greater overall recovery for the Debtor's Estate than would be provided by any other available alternative. The Debtor's determination that the Asset Purchase Agreement constitutes the highest and best offer for the Purchased Assets constitutes a valid and sound exercise of the Debtor's business judgment. The Asset Purchase Agreement represents a fair and reasonable offer to purchase the Purchased Assets under the circumstances of this Chapter 11

Case 1:15-bk-10003-MFW Doc 394 Filed 12/01/15 Entered 12/01/15 12:35:34 Desc Main Document Page 8 of 35

Case. As qualified by the Court's findings made on the record at the Sale Hearing, no other Person or entity or group of entities has offered to purchase the Purchased Assets for greater economic value to the Debtor's Estate than the Buyer. Approval of the Asset Purchase Agreement and the relief requested in the Motion, and the prompt consummation of the Transactions contemplated thereby, is in the best interests of the Debtor, its creditors, its Estate and other parties-in-interest.

No Fraudulent Transfer; Not a Successor

P. The Asset Purchase Agreement and Transaction Documents were not entered into, and the Transactions are not being consummated, for the purpose of hindering, delaying or defrauding creditors of the Debtor under applicable Law, and none of the parties to the Asset Purchase Agreement or any of the Transaction Documents are consummating the Transactions with any fraudulent or otherwise improper purpose. The Purchase Price for the Purchased Assets constitutes (i) reasonably equivalent value under the Bankruptcy Code and the Uniform Fraudulent Transfer Act, (ii) fair consideration under the Uniform Fraudulent Conveyance Act and (iii) reasonably equivalent value, fair consideration, and fair value under any other applicable Laws of the United States, any state, territory or possession of the United States or the District of Columbia.

Q. Except as expressly set forth in the Asset Purchase Agreement with respect to the Assumed Liabilities, the Buyer shall have no liability, responsibility, or obligations of any kind or nature whatsoever for any Interest of or against the Debtor, or otherwise related to the Purchased Assets, by reason of the transfer of the Purchased Assets to the Buyer. The Buyer shall not be deemed, as a result of any action taken in connection with the Transactions, to:

(1) be a successor (or other such similarly situated party) to the Debtor (other than with respect to the Assumed Liabilities as expressly stated in the Asset Purchase Agreement); or (2) have, de

Case 1:15-bk-10003-MFW Doc 394 Filed 12/01/15 Entered 12/01/15 12:35:34 Desc Main Document Page 9 of 35

facto or otherwise, merged or consolidated with or into the Debtor. The Buyer is not acquiring or assuming any Interest, except as expressly set forth in the Asset Purchase Agreement with respect to the Assumed Liabilities.

Validity of Transfer

R. Subject to the entry of this Sale Order, the Debtor has full corporate power and authority (i) to perform all of its obligations under the Asset Purchase Agreement and the Transaction Documents, and the Debtor's prior execution and delivery thereof and performance thereunder, is hereby ratified in full, and (ii) to consummate the Transactions. The Asset Purchase Agreement and Transaction Documents, and the Transactions contemplated thereby, have been duly and validly authorized by all necessary corporate action. No further consents or approvals are required for the Debtor to consummate the Transactions or otherwise perform its obligations under the Asset Purchase Agreement or the Transaction Documents, except in each case as otherwise expressly set forth in the Asset Purchase Agreement or applicable Transaction Documents.

S. As of the Closing Date, the transfer of the Purchased Assets to the Buyer, including, without limitation, the assumption, assignment and transfer of the Assigned Contracts, will be a legal, valid, and effective transfer thereof, and vests the Buyer with all right, title, and interest of the Debtor in and to the Purchased Assets, free and clear of all Interests accruing or arising any time prior to the Closing Date, except as expressly set forth in the Asset Purchase Agreement with respect to the Assumed Liabilities or Permitted Liens.

Section 363(f) Is Satisfied

T. The Buyer would not have entered into the Asset Purchase Agreement and would not consummate the Transactions contemplated thereby if the sale of the Purchased Assets, including the assumption, assignment and transfer of the Assigned Contracts, to the Buyer were not free

Case 1:15-bk-10003-MFW Doc 394 Filed 12/01/15 Entered 12/01/15 12:35:34 Desc Main Document Page 10 of 35

and clear of all Interests of any kind or nature whatsoever (except as expressly set forth in the Asset Purchase Agreement with respect to the Permitted Liens and Assumed Liabilities), or if the Buyer, any of its Affiliates or Subsidiaries, or any of their respective Representatives, would, or in the future could, be liable for any of such Interests (except as expressly set forth in the Asset Purchase Agreement with respect to the Permitted Liens and Assumed Liabilities).

U. The Debtor may sell or otherwise transfer the Purchased Assets free and clear of all Interests because, in each case, one or more of the standards set forth in section 363(f)(1)–(5) of the Bankruptcy Code has been satisfied. Those holders of Interests against the Debtor, its Estate or any of the Purchased Assets who did not object, or who withdrew their objections, to the Sale or the Motion are deemed to have consented thereto pursuant to section 363(f)(2) of the Bankruptcy Code. Those holders of such Interests who did object fall within one or more of the other subsections of section 363(f) and are adequately protected by having their Interests, if any, attach to the proceeds of the Sale ultimately attributable to the Purchased Assets in which such creditor alleges or asserts an Interest, in the same order of priority, with the same validity, force and effect, that such creditor had immediately prior to consummation of the Sale, subject to any claims and defenses the Debtor and its Estate may possess with respect thereto.

V. As used in this Sale Order, the term "Interest" includes, in addition to the types of claims described in paragraph W below, all of the following, in each case to the extent against or with respect to the Debtor or in, on, or against or with respect to any of the Purchased Assets: liens (as defined in section 101(37) of the Bankruptcy Code, and whether consensual, statutory, possessory, judicial or otherwise), claims (as defined in section 101(5) of the Bankruptcy Code), debts (as defined in section 101(12) of the Bankruptcy Code), encumbrances, obligations, Liabilities, demands, guarantees, actions, suits, defenses, deposits, credits, allowances, options, rights, restrictions, limitations, contractual commitments, rights of first refusal, rights of setoff or

Case 1:15-bk-10003-MFW Doc 394 Filed 12/01/15 Entered 12/01/15 12:35:34 Desc Main Document Page 11 of 35

recoupment, or interests of any kind or nature whatsoever, whether known or unknown, legal or equitable, matured or unmatured, contingent or noncontingent, liquidated or unliquidated, asserted or unasserted, whether arising prior to or subsequent to the commencement of this Chapter 11 Case, whether imposed by agreement, understanding, Law, equity or otherwise, including, but not limited to, (i) Interests that purport to give to any Person a right or option to effect a setoff against or any forfeiture, modification or termination of the Debtor's interests in the Purchased Assets, or any similar rights; (ii) Interests arising under all mortgages, deeds of trust, security interests, conditional sale or other title retention agreements, pledges, liens, judgments, demands, encumbrances, rights of first refusal or charges of any kind or nature, (iii) Interests that are or constitute, or that arise in connection with or with respect to, any Excluded Liability; (iv) Interests that arise from or in connection with any bulk sales or similar law, and (v) Interests arising under or in connection with any acts, or failures to act, of the Debtor or any of the Debtor's predecessors, Affiliates, or Subsidiaries, or any of their respective Representatives, including, but not limited to, Interests arising under any doctrines of successor, transferee, or vicarious liability, violation of the Securities Act of 1933, the Securities Exchange Act of 1934, or other applicable securities laws or regulations, breach of fiduciary duty, or aiding or abetting breach of fiduciary duty, or any similar theories under applicable Law or otherwise.

W. Except as expressly set forth in the Asset Purchase Agreement with respect to the Assumed Liabilities or Permitted Liens, and without limiting the nature or scope of <u>paragraph V</u> above, the transfer of the Purchased Assets, including the assumption, assignment and/or transfer of the Assigned Contracts, to the Buyer will not subject the Buyer, or its Affiliates or Subsidiaries, or any of their respective Representatives to, or subject any Purchased Asset to or provide recourse for, any Liability or Lien whatsoever with respect to the operation or condition of the Debtor's business or any of the Purchased Assets prior to the Closing or with respect to

Case 1:15-bk-10003-MFW Doc 394 Filed 12/01/15 Entered 12/01/15 12:35:34 Desc Main Document Page 12 of 35

any facts, acts, actions, omissions, circumstances or conditions existing, occurring or accruing with respect thereto prior to the Closing Date, including, without limitation, any Liability or Lien arising from any of the following: (i) any employment or labor agreements, consulting agreements, severance arrangements, change in control agreements or other similar agreements to which the Debtor is or was a party, (ii) any pension, welfare, compensation or other employee benefit plans, agreements, practices, and programs, including without limitation, any pension plan of the Debtor, (iii) the cessation of the Debtor's operations, dismissal of employees, or termination of employment or labor agreements or pension, welfare, compensation or other employee benefit plans, agreements, practices and programs and any obligations with respect thereto that arise from the Employee Retirement Income Security Act of 1974, the Fair Labor Standard Act, Title VII of the Civil rights Act of 1964, the Age Discrimination and Employment Act of 1967, the Americans with Disabilities Act of 1990, the Federal Rehabilitation Act of 1973, the National Labor Relations Act, the Virgin Islands Plant Closing Act, the Consolidated Omnibus Budget Reconciliation Act of 1985 or the Worker Adjustment and Retraining Notification Act, (iv) workmen's compensation, occupational disease or unemployment or temporary disability insurance claims, (v) environment liabilities, debts, claims or obligations which may be asserted on any basis, including, without limitation, under the Comprehensive Environmental Response, Compensation and Liability Act or any Environmental Laws, (vi) products liability or warranties, (vii) any bulk sales or similar law, (viii) any litigation by or against the Debtor and (ix) the Laws of the United States, any state, territory or possession thereof, or the District of Columbia, based, in whole or in part, directly or indirectly, in any theory of law or equity, including, without limitation, any theory of antitrust, products liability, or successor, vicarious or transferee liability. For the avoidance of doubt, the Liabilities and

Case 1:15-bk-10003-MFW Doc 394 Filed 12/01/15 Entered 12/01/15 12:35:34 Desc Main Document Page 13 of 35

Liens set forth in this paragraph are included in the defined term "<u>Interests</u>" for all purposes of this Sale Order.

Assumption, Assignment and/or Transfer of the Assigned Contracts

X. The assumption, assignment and/or transfer of the Assigned Contracts to the Buyer pursuant to the terms of this Sale Order is integral to the Asset Purchase Agreement and is in the best interests of the Debtor and its Estate, creditors and other parties in interest, and represents the reasonable exercise of sound and prudent business judgment by the Debtor.

Y. To the extent necessary or required by applicable Law, the Debtor has or will have as of the Closing Date: (i) cured, or provided adequate assurance of cure, of any default existing prior to the Closing Date with respect to the Assigned Contracts, within the meaning of sections 365(b)(1)(A) and 365(f)(2)(A) of the Bankruptcy Code, and (ii) provided compensation, or adequate assurance of compensation, to any party for any actual pecuniary loss to such party resulting from such default, within the meaning of section 365(b)(1)(B) of the Bankruptcy Code. The respective amounts set forth on Exhibit A annexed to the Cure Notices served in accordance with the Assumption and Assignment Procedures) are the sole amounts necessary under sections 365(b)(1)(A) and 365(f)(2)(A) of the Bankruptcy Code to cure all such monetary defaults and pay all actual pecuniary losses under the Assigned Contracts.

Z. The promise of the Buyer to perform the obligations first arising under the Assigned Contracts after their assumption and assignment to the Buyer constitutes adequate assurance of future performance within the meaning of sections 365(b)(1)(C) and 365(f)(2)(B) of the Bankruptcy Code to the extent that any such assurance is required and not waived by the counterparties to such Assigned Contracts. Those non-Debtor parties to Assigned Contracts who did not object to the assumption, assignment or transfer of their applicable Assigned Contract, or

Case 1:15-bk-10003-MFW Doc 394 Filed 12/01/15 Entered 12/01/15 12:35:34 Desc Main Document Page 14 of 35

to their applicable Cure Amount, are deemed to have consented thereto for all purposes of this Sale Order.

AA. Pursuant to section 2.5 of the Asset Purchase Agreement, the Buyer shall maintain certain rights to modify the list of the Assigned Contracts, after the date of this Sale Order and up to the applicable Designation Deadline as set forth in such section. Such modification rights include, but are not limited to, the right of the Buyer, prior to the applicable Designation Deadline, to designate a Contract for assumption by the Debtor and assignment to the Buyer, as well as for rejection by the Debtor. The Buyer would not have agreed to the Transactions without such modification rights. The notice and opportunity to object provided to Contract Counterparties to such Assigned Contracts and to other parties in interest, as set forth in the Assumption and Assignment Procedures contained in the Bidding Procedures Order, fairly and reasonably protects any rights that such counterparties and other parties in interest may have with respect to such Contracts.

Compelling Circumstances for an Immediate Sale

- BB. To maximize the value of the Purchased Assets and preserve the viability of the business to which the Purchased Assets relate, and to reduce the amount of postpetition debtor-in-possession financing borne by the Debtor, it is essential that the Sale of the Purchased Assets occur within the time constraints set forth in the Asset Purchase Agreement. Time is of the essence in consummating the Sale. Accordingly, there is cause to lift the stay contemplated by Bankruptcy Rules 6004 and 6006 with regards to the Transactions contemplated by this Sale Order, the Asset Purchase Agreement and the Transaction Documents.
- CC. Given all of the circumstances of this Chapter 11 Case and the adequacy and fair value of the Purchase Price under the Asset Purchase Agreement, the proposed transfer of the Purchased Assets to the Buyer constitutes a reasonable and sound exercise of the Debtor's

Case 1:15-bk-10003-MFW Doc 394 Filed 12/01/15 Entered 12/01/15 12:35:34 Desc Main Document Page 15 of 35

business judgment, is in the best interests of the Debtor, its Estate, and its creditors, and should be approved.

DD. The consummation of the Transactions is legal, valid and properly authorized under all applicable provisions of the Bankruptcy Code, including, without limitation, sections 105, 363 and 365 of the Bankruptcy Code, and all of the applicable requirements of such sections have been complied with in respect of the Transactions.

EE. The sale of Avoidance Actions described in section 2.1(a)(x) of the Asset Purchase Agreement was a material inducement for, and an express condition of, the Buyer's willingness to enter into the Asset Purchase Agreement, and when viewed with the Transactions as a whole, will provide a greater benefit to the Debtor, its Estate, and its creditors than would the retention and prosecution of such Avoidance Actions in the absence of the Transactions.

FF. The Sale does not constitute a *de facto* plan of reorganization or liquidation or an element of such a plan for the Debtor, as it does not and does not propose to: (i) impair or restructure existing debt of, or equity interests in, the Debtor; (ii) impair or circumvent voting rights with respect to any future plan proposed by the Debtor; (iii) circumvent chapter 11 plan safeguards, such as those set forth in sections 1125 and 1129 of the Bankruptcy Code; or (iv) classify claims or equity interests, compromise controversies or extend debt maturities.

NOW, THEREFORE, IT IS HEREBY ORDERED, ADJUDGED AND DECREED THAT:

General Provisions

1. The Motion, and the relief requested therein, are granted and approved, and the Transactions contemplated thereby and by the Asset Purchase Agreement and Transaction Documents are approved, in each case as set forth in this Sale Order.

- 2. This Court's findings of fact and conclusions of law set forth in the Bidding Procedures Order and the record of the Bidding Procedures Hearing are incorporated herein by reference.
- 3. All objections to the Motion or the relief requested therein that have not been withdrawn, waived, resolved, or otherwise settled as set forth herein, as announced to this Court at the Sale Hearing or by stipulation filed with this Court, and all reservations of rights included therein, are hereby denied and overruled on the merits.

Approval of Asset Purchase Agreement; Binding Nature

- 4. The Asset Purchase Agreement and the Transaction Documents, and all of the terms and conditions thereof, are hereby approved as set forth herein.
- 5. The consideration provided by the Buyer for the Purchased Assets under the Asset Purchase Agreement is fair and reasonable and shall be deemed for all purposes to constitute reasonably equivalent value, fair value, and fair consideration under the Bankruptcy Code and any other applicable Law, and the Transactions may not be avoided, or costs or damages imposed or awarded, under section 363(n) or any other provision of the Bankruptcy Code.
- 6. Pursuant to sections 363 and 365 of the Bankruptcy Code, the Debtor is authorized and empowered to, and shall, take any and all actions necessary or appropriate to (a) consummate the Sale and the other Transactions pursuant to and in accordance with the terms and conditions of the Asset Purchase Agreement and the Transaction Documents (including, without limitation, any and all actions and covenants that arise, or otherwise apply to the time period, after the Closing), and (b) execute and deliver, perform under, consummate, implement, and take any and all other acts or actions as may be reasonably necessary or appropriate to the performance of its obligations as contemplated by the Asset Purchase Agreement and the Transaction Documents, in each case without further notice to or order of this Court.

Case 1:15-bk-10003-MFW Doc 394 Filed 12/01/15 Entered 12/01/15 12:35:34 Desc Main Document Page 17 of 35

7. The Asset Purchase Agreement and Transaction Documents shall be binding in all respects upon the Debtor, its Estate, the Buyer, and all successors and assigns of the foregoing, including, without limitation, any trustee, if any, subsequently appointed in this Chapter 11 Case or upon a conversion to chapter 7 under the Bankruptcy Code of this Chapter 11 Case. This Sale Order shall be binding in all respects upon the Debtor, its Estate, all creditors, all holders of equity interests in the Debtor, all holders of any Claim(s) (whether known or unknown) against the Debtor, any holders of Interests against, in or on all or any portion of the Purchased Assets, all non-Debtor parties to the Assigned Contracts, the Buyer, and all successors and assigns of the foregoing, including, without limitation, any trustee, if any, subsequently appointed in this Chapter 11 Case or upon a conversion to chapter 7 under the Bankruptcy Code of this Chapter 11 Case.

Transfer of Purchased Assets Free and Clear of Interests; Injunction

- 8. Pursuant to sections 105(a), 363(b), 363(f), 365(b) and 365(f) of the Bankruptcy Code, the Debtor shall transfer the Purchased Assets, including but not limited to the Assigned Contracts, to the Buyer on the Closing Date in accordance with the Asset Purchase Agreement and Transaction Documents. Upon and as of the Closing Date, such transfer shall constitute a legal, valid, binding and effective transfer of such Purchased Assets and the Buyer shall take title to and possession of such Purchased Assets free and clear of all Interests (except as expressly set forth in the Asset Purchase Agreement with respect to the Permitted Liens and Assumed Liabilities).
- 9. All such Interests shall attach solely to the proceeds of the Sale with the same validity, priority, force and effect that they now have as against the Purchased Assets, subject to any claims and defenses the Debtor and its Estate may possess with respect thereto. This Sale Order shall be effective as a determination that, on and as of the Closing, all Interests of any kind

Case 1:15-bk-10003-MFW Doc 394 Filed 12/01/15 Entered 12/01/15 12:35:34 Desc Main Document Page 18 of 35

or nature whatsoever (except as expressly set forth in the Asset Purchase Agreement with respect to the Permitted Liens and Assumed Liabilities) have been unconditionally released, discharged and terminated in, on or against the Purchased Assets. The provisions of this Sale Order authorizing and approving the transfer of the Purchased Assets free and clear of Interests shall be self-executing, and neither the Debtor nor the Buyer shall be required to execute or file releases, termination statements, assignments, consents, or other instruments in order to effectuate, consummate and implement the provisions of this Sale Order.

10. Except as expressly permitted by the Asset Purchase Agreement or this Sale Order, all Persons and entities holding Interests (other than the Permitted Liens and Assumed Liabilities) are hereby forever barred, estopped and permanently enjoined from asserting their respective Interests against the Buyer, any of its Affiliates and Subsidiaries, and any of their respective Representatives, and each of their respective property and assets, including, without limitation, the Purchased Assets. On and after the Closing Date, the Buyer shall be authorized to execute and file such documents, and to take all other actions as may be necessary, on behalf of each holder of an Interest to release, discharge and terminate such Interests in, on and against the Purchased Assets as provided for herein, as such Interests may have been recorded or may otherwise exist. On and after the Closing Date, and without limiting the foregoing, the Buyer shall be authorized to file termination statements or lien terminations in any required jurisdiction to remove any record, notice filing, or financing statement recorded to attach, perfect or otherwise notice any Interest that is extinguished or otherwise released pursuant to this Sale Order. This Sale Order constitutes authorization under all applicable jurisdictions and versions of the Uniform Commercial Code for the Buyer to file UCC termination statements with respect to all security interests in or liens on the Purchased Assets. For avoidance of doubt, notwithstanding the foregoing or any other provision of this Sale Order, nothing in this Sale

Case 1:15-bk-10003-MFW Doc 394 Filed 12/01/15 Entered 12/01/15 12:35:34 Desc Main Document Page 19 of 35

Order shall modify, abridge, impair or otherwise alter any of the rights of the Debtor's debtor-inpossession lenders under the debtor-in-possession credit agreement, and the orders of this Court entered into in connection therewith, with respect to or against the Debtor.

- 11. On and after the Closing, any Persons holding an Interest (other than a Permitted Lien or an Assumed Liability) shall execute such documents and take all other actions as may be reasonably necessary to release their respective Interests in the Purchased Assets, as such Interests may have been recorded or otherwise filed. The Buyer may, but shall not be required to, file a certified copy of this Sale Order in any filing or recording office in any federal, state, county or other territory or jurisdiction in which the Debtor is incorporated or has real or personal property, or with any other appropriate clerk or recorded with any other appropriate recorder, and such filing or recording shall be accepted and shall be sufficient to release, discharge and terminate any of the Interests as set forth in this Sale Order as of the Closing Date. All Persons and entities that are in possession of any portion of the Purchased Assets on the Closing Date shall promptly surrender possession thereof to the Buyer at the Closing.
- 12. To the extent provided by section 525 of the Bankruptcy Code, no governmental unit (as defined in section 101(27) of the Bankruptcy Code) may deny, revoke, suspend, or refuse to renew a license, permit, charter, franchise, or other similar grant relating to the operation of the Purchased Assets or the Assigned Contracts, condition such a grant to, or discriminate with respect to such a grant on account of the filing or pendency of this chapter 11 case or the consummation of the Transactions. This Sale Order is and shall be binding upon and govern the acts of all Persons and entities (including, without limitation, all filing agents, filing officers, title agents, title companies, recorders of mortgages, recorders of deeds, registrars of deeds, administrative agencies, governmental departments, and secretaries of state, federal and local officials) who may be required by operation of law, the duties of their office, or contract to

Case 1:15-bk-10003-MFW Doc 394 Filed 12/01/15 Entered 12/01/15 12:35:34 Desc Main Document Page 20 of 35

accept, file, register or otherwise record or release any documents or instruments, or who may be required to report or insure any title or state of title in or to any lease. Each of the foregoing Persons and entities shall accept for filing any and all of the documents and instruments necessary and appropriate to release, discharge and terminate any of the Interests or to otherwise consummate the Transactions contemplated by this Sale Order, the Asset Purchase Agreement or any Transaction Document.

Assigned Contracts; Cure Payments

- 13. Pursuant to sections 105(a), 363 and 365 of the Bankruptcy Code, and subject to and conditioned upon the Closing Date, the Debtor's assumption, assignment and transfer to the Buyer of the Assigned Contracts are hereby authorized and approved in full subject to the terms set forth below. The Buyer shall, on or prior to the Closing, pay or cause to be paid the Cure Amounts (or reserve the Alleged Cure Amount as set forth below) under the Assigned Contracts so that such Contracts may be assumed by the Debtor and assigned to Buyer on the Closing Date in accordance with this Sale Order, the Asset Purchase Agreement and the Transaction Documents; provided that the Buyer's payment of the Pinnacle Cure Amount shall reduce the Terminals Purchase Price to the extent set forth in Section 3.1(a) of the Asset Purchase Agreement. For the avoidance of doubt, the Pinnacle Cure Amount shall be calculated by Pinnacle and the Debtor no later than two (2) business days before the Closing and shall include all amounts incurred under the Pinnacle Contract through the Closing Date.
- 14. Upon and as of the Closing, the Debtor is authorized and empowered to, and shall, assume, assign and/or transfer each of the Assigned Contracts to the Buyer free and clear of all Interests (except as expressly set forth in the Asset Purchase Agreement with respect to the Permitted Liens and Assumed Liabilities). The payment of the applicable Cure Amounts (if any), or the reservation by the Debtor or Buyer (as applicable) of an amount of cash that is equal

Case 1:15-bk-10003-MFW Doc 394 Filed 12/01/15 Entered 12/01/15 12:35:34 Des

to the lesser of (i) the amount of any cure or other compensation asserted by the applicable Contract Counterparty as required under section 365 of the Bankruptcy Code or (ii) the amount approved by order of this Court to reserve for such payment (such lesser amount, the "Alleged Cure Claim") shall, pursuant to section 365 of the Bankruptcy Code and other applicable Law, (a) effect a cure, or provide adequate assurance of cure, of all defaults existing thereunder as of the Closing Date and (b) compensate, or provide adequate assurance of compensation, for any actual pecuniary loss to such non-Debtor party resulting from such default. Accordingly, on and as of the Closing Date, other than such payment or reservation, none of the Debtor, the Buyer, or their respective Affiliates or Subsidiaries shall have any further liabilities or obligations to the non-Debtor parties to the Assigned Contracts with respect to any additional amounts or claims (as defined in section 101(5) of the Bankruptcy Code) that arose, accrued or were incurred at any time on or prior to the Closing Date. The Buyer has provided adequate assurance of future performance under the relevant Assigned Contracts within the meaning of section 365(f) of the Bankruptcy Code.

15. To the extent any provision in any Assigned Contract assumed or assumed and assigned (as applicable) pursuant to this Sale Order (a) prohibits, restricts or conditions, or purports to prohibit, restrict or condition, such assumption or assignment (including, without limitation, any "change of control" provision), or (b) is modified, breached or terminated, or deemed modified, breached or terminated by any of the following: (i) the commencement of this Chapter 11 Case, (ii) the insolvency or financial condition of the Debtor at any time before the closing of this Chapter 11 Case, (iii) the Debtor's assumption or assumption and assignment (as applicable) of such Assigned Contract, or (iv) the consummation of the Transactions, then such provision shall be deemed modified so as to not entitle the non-Debtor party thereto to prohibit, restrict or condition such assumption or assignment, to modify, terminate or declare a breach or

Case 1:15-bk-10003-MFW Doc 394 Filed 12/01/15 Entered 12/01/15 12:35:34 Desc Main Document Page 22 of 35

default under such Assigned Contract, or to exercise any other default-related rights or remedies with respect thereto, including, without limitation, any such provision that purports to allow the non-Debtor party thereto to recapture such Assigned Contracts, impose any penalty thereunder, condition any renewal or extension thereof, impose any rent acceleration or assignment fee, or increase or otherwise impose any other fees or other charges in connection therewith. All such provisions constitute unenforceable anti-assignment provisions that are void and of no force and effect pursuant to sections 365(b), 365(e) and 365(f) of the Bankruptcy Code.

- 16. All requirements and conditions under sections 363 and 365 of the Bankruptcy Code for the assumption by the Debtor and assignment to the Buyer of the Assigned Contracts have been satisfied. Upon the Closing, in accordance with sections 363 and 365 of the Bankruptcy Code, the Buyer shall be fully and irrevocably vested with all right, title and interest of the Debtor in and under the Assigned Contracts, and each Assigned Contract shall be fully enforceable by the Buyer in accordance with its respective terms and conditions, except as limited or modified by the provisions of this Sale Order. Upon and as of the Closing, the Buyer shall be deemed to be substituted for the Debtor as a party to the applicable Assigned Contracts and, accordingly, the Debtor shall be relieved from any further liability under the Assigned Contracts pursuant to section 365(k) of the Bankruptcy Code.
- 17. Upon the payment of the applicable Cure Amount or reservation of the Alleged Cure Claim, if any, the Assigned Contracts will remain in full force and effect, and no default shall exist, or be deemed to exist, under the Assigned Contracts as of the Closing Date nor shall there exist, or be deemed to exist, any event or condition which, with the passage of time or giving of notice, or both, would constitute such a default.
- 18. The rights of the Buyer to modify the list of the Assigned Contracts after the date of this Sale Order and up to the earlier of (a) the applicable Designation Deadline as set forth in

Case 1:15-bk-10003-MFW Doc 394 Filed 12/01/15 Entered 12/01/15 12:35:34 Desc Main Document Page 23 of 35

section 2.5 of the Asset Purchase Agreement, and (b) any applicable deadline under the Bankruptcy Code are hereby approved.

- 19. All Contract Counterparties to the Assigned Contracts that failed to timely object to entry of this Sale Order shall be deemed to have consented to such assumption and assignment under section 365(c)(1)(B) of the Bankruptcy Code, and the Buyer shall enjoy all of the Debtor's rights, benefits and privileges under each such Assigned Contract as of the applicable date of assumption and assignment without the necessity to obtain any non-Debtor parties' written consent to the assumption or assignment thereof.
- 20. Nothing in this Sale Order, the Motion, or in any notice or any other document is or shall be deemed an admission by the Debtor that any Assigned Contract is an executory contract or unexpired lease under section 365 of the Bankruptcy Code. The failure of the Debtor or the Buyer to enforce at any time one or more terms or conditions of any Assigned Contract shall not be a waiver of such terms or conditions, or of its respective rights to enforce every term and condition of the Assigned Contracts.

Additional Injunction; No Successor Liability

21. Effective upon the Closing Date and except as expressly set forth in the Asset Purchase Agreement with respect to the Permitted Liens and Assumed Liabilities, all Persons and entities are forever prohibited and permanently enjoined from (i) commencing or continuing in any manner any action or other proceeding, the employment of process, or any act (whether in law or equity, in any judicial, administrative, arbitral or other proceeding), to collect, recover or offset any Interest; (ii) enforcing, attaching, collecting or recovering in any manner any judgment, award, decree or order with respect to an Interest, (iii) creating, perfecting or enforcing any Interest, or (iv) asserting any setoff, right of subrogation or recoupment of any kind with respect to an Interest, in each case as against the Buyer, any of its Affiliates or

Case 1:15-bk-10003-MFW Doc 394 Filed 12/01/15 Entered 12/01/15 12:35:34 Desc Main Document Page 24 of 35

Subsidiaries, or any of their respective Representatives, or any of their respective property or assets, including the Purchased Assets.

- 22. The Transactions contemplated by the Asset Purchase Agreement and the Transaction Documents do not cause there to be, and there is not (i) a consolidation, merger, or *de facto* merger of the Buyer, on the one hand, with or into the Debtor or the Debtor's Estate, on the other hand, or vice versa; (ii) a substantial continuity between the Buyer, on the one hand, and the Debtor or the Debtor's Estate, on the other hand, (iii) a common identity between the Buyer, on the one hand, and the Debtor or the Debtor's Estate, on the other hand, or (iv) a mere continuation of the Debtor or its Estate, on the one hand, with the Buyer, on the other hand.
- 23. Except as expressly set forth in the Asset Purchase Agreement with respect to the Permitted Liens and Assumed Liabilities, the transfer of the Purchased Assets, including, without limitation, the assumption, assignment and transfer of any Assigned Contract, to the Buyer shall not cause or result in, or be deemed to cause or result in, the Buyer, any of its Affiliates or Subsidiaries, or any of their respective Representatives, having any liability, obligation, or responsibility for, or any Purchased Assets being subject to or being recourse for, any Interest whatsoever, whether arising under any doctrines of successor, transferee or vicarious liability, breach of fiduciary duty, aiding or abetting breach of fiduciary duty or otherwise, whether at Law or in equity, directly or indirectly, and whether by payment, setoff, recoupment, or otherwise.

Good Faith

24. The Transactions contemplated by this Sale Order, the Asset Purchase Agreement and Transaction Documents are undertaken by the Buyer without collusion and in good faith, as that term is defined in section 363(m) of the Bankruptcy Code, and accordingly, the reversal or modification on appeal of the authorization provided herein to consummate the Sale and other

Case 1:15-bk-10003-MFW Doc 394 Filed 12/01/15 Entered 12/01/15 12:35:34 Desc Main Document Page 25 of 35

Transactions shall not alter, affect, limit, or otherwise impair the validity of the Sale or such other Transactions (including the assumption, assignment and/or transfer of the Assigned Contracts), unless such authorization and consummation are duly stayed pending such appeal. The Buyer is a good faith purchaser within the meaning of section 363(m) of the Bankruptcy Code and, as such, is entitled to, and hereby granted, the full rights, benefits, privileges and protections of section 363(m) of the Bankruptcy Code. The purchase price reflected in the Asset Purchase Agreement was not controlled by an agreement between potential or actual bidders within the meaning of section 363(n) of the Bankruptcy Code. The Debtor and the Buyer have not engaged in any conduct that would cause or permit the Asset Purchase Agreement or the consummation of the Transactions to be avoided, or costs or damages to be imposed, under section 363(n) of the Bankruptcy Code.

Other Provisions

25. Nothing in this Sale Order or the Asset Purchase Agreement releases, nullifies, precludes, or enjoins the enforcement of any environmental liability to a governmental unit under police or regulatory statutes or regulations that any entity would otherwise be subject to as the owner or operator of property after the date of entry of this Sale Order. As relates to environmental matters, nothing in this Sale Order or the Asset Purchase Agreement authorizes (a) the transfer of any governmental license, permit, registration, authorization or approval without compliance with all applicable legal requirements under non-bankruptcy law or (b) the discontinuation of any obligation of the Debtor under any governmental license, permit, registration, authorization, or approval without compliance with all applicable legal requirements under non-bankruptcy law. Nothing in this Sale Order divests any tribunal of any jurisdiction it may have under applicable environmental law or other governmental regulatory non-bankruptcy law to interpret this Sale Order or to adjudicate any defense asserted under this Sale Order.

Case 1:15-bk-10003-MFW Doc 394 Filed 12/01/15 Entered 12/01/15 12:35:34 Desc Main Document Page 26 of 35

Nothing in this Sale Order or the Asset Purchase Agreement modifies or supersedes the Debtor's obligations under the consent decree entered into by Debtor, the United States, and the Government of the Virgin Islands in United States v. Hovensa L.L.C., No. 1:11-cv-00006 (D.V.I.).

In accordance with section 3.5(f) of the Asset Purchase Agreement, at the 26. Closing, the Buyer shall deliver or cause to be delivered to the USVI Government, the USVI Government Concession Fee (which shall be an amount equal to \$100 million) by wire transfer of immediately available funds in an account designed by the USVI Government in writing to Buyer. Payment of the USVI Government Concession Fee by the Buyer shall satisfy in full all claims of the USVI Government against the Debtor, including the claims of the GVI under the First Priority Mortgage that was issued in connection with that certain settlement agreement between the Debtor and the GVI dated May 28, 2014, but excluding any environmental claims. In accordance with, and to the extent set forth in, section 3.1(a) of the Asset Purchase Agreement, the Buyer's payment of the USVI Government Concession Fee (which constitutes part of the Buyer's purchase price for the Purchased Assets) shall reduce (on a dollar-for-dollar basis) the portion of the Buyer's purchase price to be paid to Debtor at Closing for the Purchased Assets. In accordance with, and to the extent set forth in, Section 7.33 of the Asset Purchase Agreement, the Buyer shall reimburse the Debtor (or, upon the Debtor's request, pay on the Debtor's behalf) for the Debtor's reasonable and documented out-of-pocket costs and expenses incurred in connection with the winding-up of the Debtor's affairs (including conducting Remedial Actions related to Excluded Liabilities and dismantling and disposing of any Retained Refinery Assets or other Excluded Assets) (the "Wind-Up Costs") and the performance of the Debtor's obligations under the Transaction Documents, if any; provided, however, that the aggregate amount of the Wind-Up Costs to be reimbursed (or paid on the Debtor's behalf) shall

Case 1:15-bk-10003-MFW Doc 394 Filed 12/01/15 Entered 12/01/15 12:35:34 Desc Main Document Page 27 of 35

not exceed \$30 million. Notwithstanding the foregoing, in no event shall the portion of the Buyer's purchase price to be paid to the Debtor at Closing for the Purchased Assets be less than \$90 million, less the Pinnacle Cure Amount.

- 27. At the Closing, the Debtor shall repay in full in cash to the DIP Lenders (as defined in the Final Order (A) Approving Postpetition Financing, (B) Granting Liens and Providing Superpriority Administrative Expense Claims, and (C) Modifying the Automatic Stay (the "Final DIP Order") all accrued but unpaid DIP Obligations as such term is defined in the Final DIP Order. Such payment shall be in full and complete settlement, release, and discharge of all DIP Obligations.
- 28. The Buyer is hereby authorized, in its discretion, in connection with consummation of the Transactions to allocate the Purchased Assets, Assumed Liabilities, and Assigned Contracts among its Affiliates, Subsidiaries, designees, assignees, and/or successors in a manner as it, in its discretion, deems appropriate and such Person shall be entitled to all of the rights, benefits, privileges and protections of "the Buyer" as are accorded to the Buyer under this Sale Order, and the Debtor shall, to the extent set forth in the Asset Purchase Agreement and Transaction Documents, cooperate with and take all actions reasonably requested by Buyer to effectuate any of the foregoing. In the event that the Buyer designates any designee to acquire any Purchased Assets, including, without limitation, any Assigned Contracts, in accordance with the terms and conditions of the Asset Purchase Agreement, then any reference to "the Buyer" in this Sale Order shall be deemed to be a reference to "the Buyer and/or its applicable permitted designee," unless the context requires otherwise, and such permitted designee shall be entitled to, and is hereby granted, the full rights, benefits, privileges and protections of the Buyer as are set forth in this Sale Order.

Case 1:15-bk-10003-MFW Doc 394 Filed 12/01/15 Entered 12/01/15 12:35:34 Desc Main Document Page 28 of 35

- 29. To the extent any plan of reorganization or liquidation, or any order of any type or kind entered in this Chapter 11 Case or any subsequent chapter 7 case into which this Chapter 11 Case may be converted, conflicts with or derogates from the terms of the Asset Purchase Agreement or this Sale Order, the terms of the Asset Purchase Agreement and this Sale Order shall control and govern to the extent of any such conflict or derogation.
- 30. Pursuant to Bankruptcy Rules 6004(h), 6006(d), 7062, and 9014, this Sale Order shall not be stayed after the entry hereof, but shall be effective and enforceable immediately upon entry, and the stays provided in Bankruptcy Rules 6004(h) and 6006(d) are hereby expressly waived and shall not apply. Accordingly, the Debtor is authorized and empowered to close the Sale and other Transactions immediately upon entry of this Sale Order.
- 31. Notwithstanding anything contained in this Sale Order, the terms and conditions of this Sale Order shall be effective and enforceable only upon the satisfaction of each of the following additional conditions: (i) PDVSA, V.I., Inc. ("PDV-VI") shall have received approvals for the Sale and other Transactions, including with respect to any waiver of pre-petition claims held by PDV-VI and/or its Affiliates against the USVI Government, HOVIC, Hess, and/or their respective Affiliates, on or before December 14, 2015 (the "Outer Date"); provided, that such waiver shall not bar PDV-VI, HOVIC, and/or its Affiliates from asserting such claims for purposes of setoff, offset, recoupment or other defensive purposes (which claims, for the avoidance of doubt, shall not be entitled to any affirmative recovery from the Debtor); (ii) the releases contemplated by section 8.3(f) of the Asset Purchase Agreement shall have been finalized and executed on or before the End Date, in form and substance acceptable to each of the parties thereto in their sole respective capacities, which agreement shall, among other things, provide for the mutual release of claims by and among HOVIC, PDV-VI, the USVI Government, and/or their respective Affiliates; (iii) on or before the Outer Date, HOVIC or one of its

Affiliates shall have agreed to assume the Debtor's pension obligations arising from the HOVENSA Employees' Pension Plan, in a form and manner acceptable to HOVIC, the Committee, the Pension Benefit Guaranty Corporation, and the Debtor and in accordance with the requirements of the Internal Revenue Code, ERISA and applicable law, which such assumption shall be subject to and occur upon consummation of a liquidating plan that contains estate releases for HOVIC and its Affiliates; (iv) the Legislature of the U.S. Virgin Islands (the "Legislature") shall have approved the Sale and other Transactions by the End Date and such approval shall not limit or affect in any way the releases granted in clause (ii) of this Paragraph; (v) the Committee shall be satisfied with respect to the scope of the claims that may be asserted against the Debtor by the Environmental Protection Agency or shall have waived this condition on or before the Outer Date; and (vi) on or before the Outer Date: (a) the Office of the United States Trustee, the Committee, and the Debtor shall be satisfied that the letter agreement, dated November 9.2015, between the Governor of the U.S. Virgin Islands and the Buyer regarding the proposed concession agreement has no economic impact on the Debtor or its estate beyond what is already expressly included in the terms and conditions of the Asset Purchase Agreement and (b) the Buyer shall have filed the letter agreement with the Court (collectively, the "Additional Closing Conditions"); provided, however, that the Outer Date may be extended by the mutual agreement of the Debtor, the Committee, PDV-VI, HOVIC, and the USVI Government, as appropriate, in consultation with the Buyer; and further provided that, if any of the Additional Closing Conditions fail to occur by the Outer Date or the End Date, as applicable, and are not waived, the Debtor shall not seek to set a hearing date on its motion to approve the Sale to the Buyer that is less than thirty (30) days from the Outer Date, as such date may have been extended from time to time. The Debtor shall notify the Committee and the Buyer of the occurrence of the Additional Closing Conditions identified in (i), (ii) and (iii) above in writing

Case 1:15-bk-10003-MFW Doc 394 Filed 12/01/15 Entered 12/01/15 12:35:34 Desc Main Document Page 30 of 35

promptly upon the occurrence of such events, the Committee shall notify the Debtor and the Buyer of the occurrence of the Additional Closing Condition identified in (v) above in writing promptly upon the occurrence of such event, and the Buyer shall notify the Debtor of the occurrence of the Additional Closing Condition identified in (vi) above in writing promptly upon the occurrence of such event. The rights of all parties are reserved with respect to the Sale prior to satisfaction or waiver of the Additional Closing Conditions. Upon satisfaction or waiver of the Additional Closing Conditions, the Debtor shall promptly file a notice thereof with this Court and may thereafter, subject to the terms and conditions of the Asset Purchase Agreement, seek to consummate the Sale and other Transactions contemplated by the Asset Purchase Agreement without further notice to, order of, or hearing before this Court. Nothing in this paragraph, including, without limitation, the extension of the Outer Date, shall, or shall be deemed to, amend, modify, or waive any term or condition of the Asset Purchase Agreement (including any condition precedent to the Sale or the right and timing of the Buyer to receive the return of its Good Faith Deposit) or limit, alter, or impair the ability of the Buyer to terminate the Asset Purchase Agreement in accordance with the terms and conditions thereof.

- 32. Nothing in this Sale Order shall modify or waive any closing conditions or termination rights in Articles VIII and IX, respectively, of the Asset Purchase Agreement, and all such conditions and rights shall remain in full force and effect in accordance with their terms.
- 33. No bulk sales law or any similar law of any state or other jurisdiction applies in any way to the Transactions.
- 34. All payment or reimbursement obligations of the Debtor owed to the Buyer pursuant to the Asset Purchase Agreement or the Transaction Documents shall be paid in the manner provided therein, without further notice to or order of this Court. All such obligations shall constitute allowed administrative claims against the Debtor, with priority administrative

Case 1:15-bk-10003-MFW Doc 394 Filed 12/01/15 Entered 12/01/15 12:35:34 Desc Main Document Page 31 of 35

expense status under sections 503(b) and 507(a)(2) of the Bankruptcy Code, junior in priority to any super-priority administrative expense claim granted to, or on behalf of, the lenders in connection with any debtor-in-possession financing approved by Bankruptcy Court. Until satisfied in full in cash, all such obligations shall continue to have the protections provided in this Sale Order, and shall not be discharged, modified or otherwise affected by any reorganization plan for the Debtor.

- 35. The failure specifically to include any particular provision of the Asset Purchase Agreement in this Sale Order shall not diminish or impair the effectiveness of such provision, it being the intent of this Court that the Asset Purchase Agreement be authorized and approved in its entirety.
- 36. The Asset Purchase Agreement and Transaction Documents may be modified, amended or supplemented in a writing signed by the parties thereto and in accordance with the terms thereof, without further notice to or order of this Court, <u>provided</u> that any such modification, amendment or supplement shall not have a material adverse effect on the Debtor's Estate unless approved by order of this Court. All such modifications, amendments and/or supplements shall be provided promptly to counsel to the Committee.
- 37. This Court retains jurisdiction, pursuant to its statutory powers under 28 U.S.C. § 157(b), to, among other things, (i) interpret, implement, and enforce the terms and provisions of this Sale Order, the Asset Purchase Agreement, the Transaction Documents, and any amendments thereto and any waivers and consents given thereunder, (ii) compel delivery of the Purchased Assets to the Buyer; (iii) enforce the injunctions and limitations of liability set forth in this Sale Order, and (iv) enter any orders under sections 105, 363 and 365 of the Bankruptcy Code with respect to the Purchased Assets and Assigned Contracts.

Case 1:15-bk-10003-MFW Doc 394 Filed 12/01/15 Entered 12/01/15 12:35:34 Desc Main Document Page 32 of 35

- 38. All time periods set forth in this Sale Order shall be calculated in accordance with Bankruptcy Rule 9006(a). The rules of construction set forth in section 1.3 of the Asset Purchase Agreement shall apply to this Sale Order, *mutatis mutandis*.
- 39. The automatic stay provisions of section 362 of the Bankruptcy Code are vacated and modified to the extent necessary to implement the provisions of this Sale Order and the terms and conditions of the Asset Purchase Agreement and the Transaction Documents.
- 40. To the extent that this Sale Order is inconsistent with the Motion, the terms of this Sale Order shall control and govern. To the extent there are any inconsistencies between the terms of this Sale Order, on the one hand, and the Asset Purchase Agreement or any Transaction Document, on the other hand, the terms of this Sale Order shall control and govern.
- 41. Notwithstanding anything to the contrary contained in this Sale Order, any agreement or other document, upon closing of the Sale contemplated by this Sale Order, the sum of \$30 million shall be placed in a separate interest bearing account (the "UCC Beneficiary Account") for the sole benefit of the UCC Beneficiaries (as defined below), which amount shall be subject only to the UCC Carveout (as defined below). Sums from the UCC Beneficiary Account shall be distributed for UCC Beneficiaries upon and after the consummation of a chapter 11 plan of liquidation, under terms acceptable to the Committee, the Debtor, and any liquidating trustee, as appropriate. UCC Beneficiaries include holders of allowed nonpriority general unsecured claims other than: (1) claims held by HOVIC or PDV-VI; (2) any claims of the USVI Government; (3) any claim of any governmental entity, unless otherwise agreed in writing by the Committee, the Debtor, and any liquidating trustee, as appropriate; (4) any claim of any parent, affiliate, assigns, officer, director, employee or agent of (1) through (3); provided, however, nothing herein shall limit or affect in any way the rights of such officers or directors to assert claims (i) against the Debtor's insurance carrier and/or (ii) for purposes of setoff, offset,

Case 1:15-bk-10003-MFW Doc 394 Filed 12/01/15 Entered 12/01/15 12:35:34 Desc Main Document Page 33 of 35

recoupment or other defensive purposes (which claims, for the avoidance of doubt, shall not be entitled to any affirmative recovery from the Debtor). UCC Carveout means any amount in excess of the aggregate amount budgeted for the Committee, its members and Committee professionals, and for any post-plan confirmation costs and fees incurred in connection with determining the validity of UCC Beneficiary Claims and making distributions to UCC Beneficiaries under the terms acceptable to the Committee, the Debtor, and any liquidating trustee, as appropriate.

- 42. Buckeye Partners, L.P. ("<u>Buckeye</u>") is hereby designated as the Backup Bidder (as defined in the Bidding Procedures). The Debtor is authorized to close a transaction with Buckeye as Backup Bidder provided that: (a) the conditions to closing of the transaction with the Buyer (including, without limitation, approval by the Governor and Legislature of the proposed concession agreement between the USVI Government and the Buyer) are not satisfied or waived, as applicable, in accordance with the terms and conditions of the Asset Purchase Agreement; (b) the conditions to closing of the transaction with the Backup Bidder (including, without limitation, approval by the Governor and Legislature of the proposed concession agreement between the USVI Government and the Backup Bidder) are satisfied or waived, as applicable, in accordance with the terms and conditions of the asset purchase agreement governing the Backup Bid (as defined in the Bidding Procedures); and (c) Buckeye has removed the condition precedent in its bid with respect to any proceeding commenced by the Federal Trade Commission as described on the record during the Sale Hearing.
- 43. At Closing, Buyer or Back-up Bidder (as defined in the Bidding Procedures) will deposit into a segregated account or escrow account in the State of Delaware the sum of \$8,968,147.66 in Sale proceeds (the "<u>Proceeds</u>") as adequate protection for the charging lien and equitable interest of John K. Dema, P.C. (the "<u>Dema Firm</u>"). The charging lien and equitable

Case 1:15-bk-10003-MFW Doc 394 Filed 12/01/15 Entered 12/01/15 12:35:34 Desc Main Document Page 34 of 35

interest of the Dema Firm will attach to the Proceeds notwithstanding the payment to the Government of the Virgin Islands in satisfaction of the Government of the Virgin Islands' own claims under paragraph 26 of this Sale Order. Upon execution of an escrow agreement mutually acceptable to the closing Buyer or Back-up Bidder (the "Successful Purchaser") and the Dema Firm on or prior to December 16, 2015, the Proceeds will be deposited into a segregated or escrow account with an escrow agent that is mutually acceptable to the Successful Purchaser and the Dema Firm. In the absence of such an agreement on or before December 16, 2015, the Court will determine the terms of such escrow agreement including the specification of the escrow account with an escrow agent. No Proceeds will be disbursed from the segregated account or the escrow account absent (a) written agreement of the Successful Purchaser and the Dema Firm, (b) an arbitration award, or (c) an order of this Court. The Proceeds will not be subject to garnishment, execution, or other process by any (i) creditor of the Successful Purchaser, (ii) the Debtors, (iii) the Government of the Virgin Islands, or (iv) any other entity as such funds are in custodia legis for the benefit of the Dema Firm. To the extent the amount of an agreement between the Successful Purchaser and Dema Firm, an arbitration award, or order of this Court are in an amount less than \$8,968,147.66, the balance of funds remaining shall revert to the Successful Purchaser. Notwithstanding Successful Purchaser's agreement to segregate or escrow the Proceeds, this right to the balance of funds remaining, if any, is the only right Successful Purchaser has to the funds. The Successful Purchaser waives all other rights and shall not appear adverse to the Dema Firm in any proceeding, including arbitration, related to the Dema Firm's entitlement to the funds. For the avoidance of doubt, the Proceeds referred to in this paragraph are in addition to the Sale proceeds addressed in paragraph 26 of this Sale Order.

Case 1:15-bk-10003-MFW Doc 394 Filed 12/01/15 Entered 12/01/15 12:35:34 Desc Main Document Page 35 of 35

Dated: December | , 2015 St. Croix, Virgin Islands

JNITED STATES BANKRUPTCY JUDGE

EXHIBIT A

Asset Purchase Agreement

AMENDED AND RESTATED

ASSET PURCHASE AGREEMENT

BY AND AMONG

LIMETREE BAY HOLDINGS, LLC,

HOVENSA L.L.C.,

AND

HESS OIL VIRGIN ISLANDS CORP.

DATED DECEMBER [__], 2015

ARTICLE ID	EFINITIONS; CONSTRUCTION	2
1.1	Defined Terms	. 2
1.2	Additional Defined Terms	
1.3	Construction	
1.4	Exhibits and the Disclosure Letters	
1.5	Knowledge	
	SALE OF ASSETS	
0.1		
2.1	Sale of Purchased Assets and Tug Boats	
2.2	Excluded Assets	
2.3	Assumption of Liabilities	
2.4	Excluded Liabilities	
2.5	Assumption and Assignment of Contracts	
2.6	Required Consents	30
ARTICLE III	PURCHASE PRICE	31
3.1	Purchase Price; Delivery of Funds.	31
3.2	Good Faith Escrow Deposit.	
3.3	Allocation of the Purchase Price	
3.4	Withholding	
3.5	Closing; Closing Deliverables	
3.0		
ARTICLE IV	REPRESENTATIONS AND WARRANTIES OF SELLER	36
4.1	Due Organization, Good Standing and Limited Liability Company Power	36
4.2	Authorization; Noncontravention	
4.3	Governmental Consents and Approvals.	
4.4	Property	
4.5	Affiliate Transactions	
4.6	Material Contracts	
4.7	Intellectual Property Rights and Claims	
4.8	Tax Matters	
4.9	Employee Benefits	43
4.10	Compliance with Laws	
4.11	Labor Matters	
4.12	Environmental Matters	
4.13	Finders; Brokers	
4.14	Permits	
4.15	Absence of Changes.	
4.16	Litigation	
4.17	Title to Purchased Assets	
4.18	Insurance	
4.19	Compliance.	
4.20	Exclusivity of Representations; Projections, etc.	

ARTICLE V	REPRESENTATIONS AND WARRANTIES OF HOVIC	46
5.1	Due Organization, Good Standing and Corporate Power	
5.2	Authorization; Noncontravention.	
5.3	Governmental Consents and Approvals	
5.4	Litigation	48
5.5	Title to Tug Boats	
5.6	Exclusivity of Representations; Projections, etc	48
ARTICLE V	TREPRESENTATIONS AND WARRANTIES OF PURCHASER	49
6.1	Due Organization, Good Standing and Limited Liability Company Power	
	Purchaser	
6.2	Authorization; Noncontravention	
6.3	Governmental Consents and Approvals	
6.4	Financing	
6.5	Litigation	
6.6	Finders; Brokers	
6.7	Investigation by Purchaser	
6.8	Solvency of Purchaser	
6.9	Exclusivity of Representations	52
ARTICLE V	TICOVENANTS	53
7.1	Access to Information Concerning Properties and Records	53
7.2	Conduct Pending the Closing Date	54
7.3	Employee Relations and Benefits	56
7.4	Efforts to Close	
7.5	Public Announcements	
7.6	Notification of Certain Matters	
7.7	Supplements to Schedules	
7.8	Post-Closing Access to Records and Personnel; Litigation Support	
7.9	Compliance with WARN, Plant Closing Act and Similar Statutes	
7.10	Tax Matters	
7.11	USVI Concession Agreement	
7.12	Access Easement	
7.13	Bulk Sales Act	
7.14	Bankruptcy Action	
7.15	Sale Order.	
7.16	[Intentionally Omitted]	
7.17	Backup Bidder	
7.18	Transfer of Permits; Seller Letters of Credit	
7.19	Expenses	
7.20	Breakup Fee.	
7.21	Rejected Contracts	
7.22	Confidentiality	
7.23	Seller's Marks	60

Case 1:15-bk-10003-MFW Doc 394-1 Filed 12/01/15 Entered 12/01/15 12:35:34 Desc Exhibit Page 5 of 49

	7.24	Shared Services Agreement	69
	7.25	Permitting Process	
	7.26	Registration of Tug Boats	
	7.27	Post-Closing Cooperation for Financial Statements	
	7.28	Division of Subdivision Parcels; Ground Leases; Cross-Easement Agreements.	71
	7.29	Title Policy and Survey	
	7.30	Additional Cross-Easement Agreements.	
	7.31	Cooperation	
	7.32	Transfer of Government Parcels	
	7.33	Costs and Expenses of Seller	
ARTIC	CLE VII	ICONDITIONS PRECEDENT	73
	8.1	Conditions to the Obligations of Each Party	73
	8.2	Conditions to the Obligations of Purchaser.	74
	8.3	Conditions to the Obligations of Seller and HOVIC	75
	8.4	Frustration of Closing Conditions	
ARTIC	CLE IX	ΓERMINATION	76
	9.1	Termination Events	76
	9.2	Effect of Termination.	78
ARTIC	CLE XS	URVIVAL AND RELEASE	78
	10.1	No Survival of Representations and Warranties	78
	10.2	Release	
ARTIC	CLE XII	MISCELLANEOUS	79
	11.1	Expenses	79
	11.2	Extension; Waiver.	
	11.3	Notices	80
	11.4	Entire Agreement	
	11.5	Binding Effect; Benefit; Assignment	
	11.6	Amendment and Modification	
	11.7	Counterparts	
	11.8	Applicable Law; Jurisdiction	
	11.9	Severability	
	11.10	Specific Enforcement; Limitation on Damages	
	11.11	Waiver of Jury Trial.	
	11.12	Rules of Construction	
	11.13	Headings	
			85

Case 1:15-bk-10003-MFW Doc 394-1 Filed 12/01/15 Entered 12/01/15 12:35:34 Desc Exhibit Page 6 of 49

EXHIBITS

Exhibit A: Form of Access Easement Exhibit B: [Intentionally Omitted]

Exhibit C: Form of Bill of Sale and Assignment and Assumption Agreement

Exhibit D: Form of Protocol of Delivery and Acceptance Exhibit E: Form of Tug Boat Bill of Sale and Acceptance

Exhibit F: Form of USVI Concession Agreement

Exhibit G: Form of FIRPTA Certificate

Exhibit H: Subdivision Parcels

Exhibit I: Form of Termination and Release for Concession Agreement, the 1976 Contract

and the Submerged Land Lease

OTHER

Seller Disclosure Letter HOVIC Disclosure Letter

AMENDED AND RESTATED ASSET PURCHASE AGREEMENT

This AMENDED AND RESTATED ASSET PURCHASE AGREEMENT (this "Agreement") is dated December [__], 2015 by and among Limetree Bay Holdings, LLC, a limited liability company organized under the Laws of Delaware ("Purchaser"), HOVENSA L.L.C., a limited liability company organized under the Laws of the U.S. Virgin Islands ("Seller") and Hess Oil Virgin Islands Corp., a corporation organized under the Laws of the U.S. Virgin Islands ("HOVIC" and together with Purchaser and Seller, the "Parties" and each a "Party"). Capitalized terms used herein have the meanings ascribed thereto in Section 1.1, unless otherwise provided.

$\underline{W} \underline{I} \underline{T} \underline{N} \underline{E} \underline{S} \underline{S} \underline{E} \underline{T} \underline{H}$:

WHEREAS, Seller has previously been engaged, in part, in the business of owning and operating a crude oil and product storage and terminalling business (the "Business");

WHEREAS, on September 15, 2015, Seller commenced a voluntary case (the "Bankruptcy Case") under chapter 11 of the Bankruptcy Code in the United States District Court of the U.S. Virgin Islands, Bankruptcy Court Division – St. Croix, Virgin Islands (the "Bankruptcy Court");

WHEREAS, Seller retains possession of its assets and is authorized under the Bankruptcy Code to continue the operation of its business as a debtor-in-possession under sections 1107(a) and 1108 of the Bankruptcy Code;

WHEREAS, HOVIC owns the Tug Boats;

WHEREAS, in connection with the Bankruptcy Case and subject to the respective terms and conditions contained herein and in the Sale Order, Seller desires to sell and assign the Purchased Assets and the Assumed Liabilities to Purchaser, and Purchaser desires to purchase and assume the Purchased Assets and the Assumed Liabilities, pursuant to sections 105, 363 and 365 of the Bankruptcy Code, free and clear of all Liens, Claims, and Liabilities other than Permitted Liens and Assumed Liabilities, upon the terms and conditions set forth in this Agreement (the "Purchase") and HOVIC desires to sell to Purchaser, and Purchaser desires to purchase, on the terms and subject to the conditions of this Agreement, the Tug Boats (the "Tug Boat Sale");

WHEREAS, Seller has determined, in the exercise of its business judgment, that it is advisable and in the best interest of the estate and the beneficiaries of the estate to consummate the Purchase provided for herein pursuant to the Bid Procedures Order and the Sale Order;

WHEREAS, the Parties entered into an Asset Purchase Agreement, dated September 4, 2015 (the "Initial Purchase Agreement");

WHEREAS, the Parties desire to amend and restate the Initial Purchase Agreement in its entirety;

WHEREAS, this Agreement amends and restates the Initial Purchase Agreement in its entirety; and

NOW, THEREFORE, in consideration of the premises and of the mutual covenants, representations, warranties and agreements herein contained, the Parties, intending to be legally bound, agree as follows:

ARTICLE I

DEFINITIONS; CONSTRUCTION

- 1.1 <u>Defined Terms</u>. When used in this Agreement, the following terms shall have the respective meanings specified therefor below:
- "1976 Contract" shall mean that certain Contract, dated September 22, 1976, by and among the USVI Government, Seller (as assignee of HOVIC) and the Virgin Islands Port Authority, approved by the Legislature of the Virgin Islands on September 29, 1976, and including, for the avoidance of doubt, the 1976 Contract Permits.
- "1998 Letter Agreement" shall mean that certain letter agreement, dated as of October 14, 1998 (as amended, supplemented or modified from time to time pursuant to which the USVI Government consented to (a) the assignment by HOVIC to Seller of the Submerged Land Lease and (b) the assignment and delegation by HOVIC to Hovensa of the rights and obligations of HOVIC under the 1976 Contract.
- "2012 HOVIC Note" shall mean that certain note issued as of April 1, 2012 by Seller in favor of HOVIC in the aggregate original principal amount of \$811,000,000.
 - "2012 Notes" shall mean the 2012 HOVIC Note and the 2012 Petróleo Note.
- "2012 Petróleo Note" shall mean that certain note issued as of April 1, 2012 by Seller in favor of PDVSA Petróleo, S.A. in the aggregate original principal amount of \$811,000,000.
- "2015 HOVIC Note" shall mean that certain note issued as of July 8, 2015 by Hovensa in favor of HOVIC in the aggregate original principal amount of \$5,000,000.
 - "2015 Notes" shall mean the 2015 HOVIC Note and the 2015 PDVSA Note.

"2015 PDVSA Note" shall mean that certain note issued as of July 8, 2015 by Hovensa in favor of PDVSA VI in the aggregate original principal amount of \$5,000,000.

"Above-Grade Refinery Assets" shall mean any and all refining process units, buildings, structures, fixtures or other improvements owned by Seller that are present on the Option Refinery Property or the Leased Submerged Lands [or that certain parcel of real property identified in Exhibit H as an "Excluded" parcel and located within the submerged land border] that are exclusively at or above grade, including the power plant located on the Option Refinery Property, and all equipment, personal property and fixtures, including any and all ancillary and non-structural personal property, required to operate such assets (including below-grade pumps, storage tanks, piping, electrical service and distribution system, control systems and any other associated utility infrastructure); provided, that, for the avoidance of doubt, the Above-Grade Refinery Assets shall not include any Retained Refinery Assets or other Excluded Assets.

"Access Easement" shall mean an access easement in form and substance reasonably satisfactory to each of Seller and Purchaser on terms substantially similar to the form attached hereto as Exhibit A.

"Adverse Person" shall mean (a) a Person who is a "designated national," "specially designated national," "specially designated terrorist," "specially designated global terrorist," "foreign terrorist organization," "specially designated narcotics trafficker" or "blocked person," within the definitions set forth in the OFAC Regulations, or who otherwise appears on the list of Specially Designated Nationals and Blocked Persons included in the OFAC Regulations, (b) a government, including any Governmental Entity, of any country against which the United States maintains Sanctions, (c) a Person acting or purporting to act, directly or indirectly, on behalf of, or any entity owned or controlled by, any of the Persons listed in clauses (a) or (b) above, or (d) a Person on any other export control, terrorism or drug trafficking related list administered by any Governmental Entity as that list may be amended, adjusted or modified from time to time.

"Affiliate" of any Person shall mean any other Person directly or indirectly controlling, controlled by, or under common control with, such Person; provided, that, for the purposes of this definition, "control" (including, with correlative meanings, the terms "controlled by" and "under common control with"), as used with respect to any Person, shall mean the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, whether through the ownership of voting securities, by Contract or otherwise; provided, further, that none of HOVIC, PDVSA VI or their respective direct or indirect equity holders shall be deemed to "control" Seller. Notwithstanding the foregoing, for all purposes of this Agreement and the transactions contemplated hereby, (a) HOVIC and its direct and indirect equity holders are Affiliates of Seller but are not Affiliates of PDVSA VI or PDVSA VI's direct and indirect equity holders are Affiliates of Seller but are not Affiliates of HOVIC or HOVIC's direct and indirect equity holders.

Note to Draft: Subject to confirmation with Purchaser.

- "Alternative Transaction" shall mean (i) a Restructuring Transaction or (ii) one or more sales, assignments, leases, transfers, or other dispositions of all or any material portion of the Purchased Assets to any Person (or group of Persons), whether in one transaction or a series of transactions, whether by merger, asset purchase, equity purchase or other similar transaction, in each case other than to the Purchaser or its Affiliates.
- "Antitrust Laws" shall mean the Sherman Act, 15 U.S.C. §§ 1-7, as amended, the Clayton Act, 15 U.S.C. §§ 12-27, 29 U.S.C. §§ 52-53, as amended, the Hart-Scott-Rodino Antitrust Improvements Act of 1976, 15 U.S.C. § 18a et seq., as amended, the Federal Trade Commission Act, 15 U.S.C. § 41-58, as amended, and all other Laws and Orders that are designed or intended to prohibit, restrict or regulate actions having the purpose or effect of monopolization, restraint of trade, or lessening of competition through merger or acquisition.
- "Avoidance Actions" shall mean all preference claims, fraudulent transfer claims and avoidance actions or other related causes of action whether arising under the Bankruptcy Code, non-bankruptcy law, or otherwise, and the proceeds thereof, including actions available to Seller under Chapter 5 of the Bankruptcy Code, of whatever kind or nature, and whether asserted or unasserted.
- "Bankruptcy Code" shall mean Title 11 of the United States Code, 11 U.S.C. §§ 101, et seq.
- "Bid Procedures Order" shall mean the Order (a) Establishing Bidding Procedures Relating to the Sale of the Debtor's Assets, Including Approving Break-Up Fee and Expense Reimbursement, (b) Establishing Procedures Relating to the Assumption and Assignment of Certain Executory Contracts and Unexpired Leases, Including Notice of Proposed Cure Amounts, (c) Approving Form and Manner of Notice Relating Thereto, and (d) Scheduling a Hearing to Consider the Proposed Sale entered by the Bankruptcy Court on October 9, 2015.
- "Bill of Sale and Assignment and Assumption Agreement" shall mean that certain bill of sale and assignment and assumption agreement substantially in the form attached hereto as Exhibit C.
 - "Breakup Fee" shall mean an amount of cash equal to \$4,700,000.
- "Business Day" shall mean any day except a Saturday, a Sunday or any other day on which commercial banks are required or authorized to close in New York, New York or the U.S. Virgin Islands.
- "Business Intellectual Property" shall mean all rights to Intellectual Property that are owned by Seller, other than the Excluded Intellectual Property.
- "Business IT Assets" shall mean all rights to Information Technology that are owned by Seller, other than the Excluded IT Assets.
- "Business Real Property" shall mean the Purchased Real Property and the Leased Submerged Lands, in each case together with all easements, appurtenances, rights and

leases, and other hereditaments appurtenant to such land and all the estates and rights of Seller in and to such land.

"Charter Agreements" shall mean, with respect to each Tug Boat, the applicable Contract of Bareboat Charter Party, dated as of October 30, 1998, by and between Amerada Hess Shipping Corporation and Seller, as modified by the applicable Novation Agreement, dated as of July 20, 2001, by and among Amerada Hess Shipping Corporation, Seller and HOVIC.

"Claim" shall have the meaning assigned to such term under section 101(5) of the Bankruptcy Code.

"Code" shall mean the United States Internal Revenue Code of 1986, as amended, and the regulations promulgated thereunder, as mirrored in the U.S. Virgin Islands.

"Concession Agreement" shall mean that certain agreement by and between the USVI Government and HOVIC, dated and approved by the Legislature of the Virgin Islands September 1, 1965, and amended, supplemented and clarified at various times by mutual agreement of the parties, as amended and extended by the Extension and Amendment Agreement, dated April 24, 1981 and approved by the Legislature of the Virgin Islands May 7, 1981, as further amended and extended by the Restated Second Extension and Amendment Agreement, dated July 27, 1990 and approved by the Legislature of the Virgin Islands on August 22, 1990, as further amended by the Technical Clarifying Amendment to Restated Second Extension and Amendment Agreement, dated November 17, 1993 and approved by the Governor and the Legislature of the Virgin Islands, as further amended and extended by the Third Extension and Amendment Agreement, to which PDVSA VI is added as a party, dated April 15, 1998 and approved by the Legislature of the Virgin Islands on May 18, 1998, as further amended by the Fourth Amendment Agreement.

"Confidentiality Agreement" shall mean that certain Confidentiality Agreement by and between Seller and ArcLight Capital Partners, LLC, dated April 24, 2015.

"Consent" shall mean any authorization, approval, consent, ratification, negative clearance, waiver, notice or filing, or a Final Order of the Bankruptcy Court that deems, or renders unnecessary, the same.

"Consent Decree" shall mean the Consent Decree entered on June 7, 2011 in the United States District Court of the Virgin Islands in the matter of United States of America, and United States Virgin Islands v. Hovensa L.L.C. (Civ. No.: 1:11-cv-00006).

"Contract" shall mean any note, bond, mortgage, indenture, guaranty, license, franchise, permit, agreement, contract, commitment, lease, purchase order, or other instrument or obligation, and any amendments thereto.

"Cross-Easement Agreement" shall mean an agreement, in form and substance reasonably acceptable to each of Seller and Purchaser, by which each of Seller and Purchaser grants the other a customary right of access upon, over, in, under, across and through a portion of its property as is reasonably required for the continued use and operation of the Purchased Real Property, the leased real property subject to any Real Property Leases (including, for the

avoidance of doubt, the Leased Submerged Lands) or the Excluded Real Property by Purchaser or Seller (as applicable).

"Cure Amounts" shall mean, with respect to any Business Contract or Real Property Lease, the amount of cash required to be paid with respect to such Business Contract or Real Property Lease to cure all defaults under such Business Contract or Real Property Lease to the extent required by Section 365 of the Bankruptcy Code and to otherwise satisfy all requirements imposed by Section 365 of the Bankruptcy Code in order to effectuate, pursuant to the Bankruptcy Code, the assumption by Seller and assignment to Purchaser of such Business Contract or Real Property Lease.

"Disclosure Conditions" shall mean, in respect of any obligation of a Person to disclose any information or provide access to another Person, that such disclosing Person shall not be required to disclose such information or provide such access to the extent that such Person reasonably believes that doing so would reasonably be expected to (a) violate any applicable Law or Order, including Antitrust Laws, or the terms of any Contract to which such Person is a party, (b) cause the waiver of attorney/client privilege or similar privilege or (c) unreasonably disrupt the operations of such Person; provided that to the extent permitted by Law, the Person that has the disclosure obligation shall notify the other Person of any such inability to provide information or access and shall use its commercially reasonable efforts to make appropriate substitute arrangements (including entering into joint defense agreements, seeking appropriate waivers or consents) under the circumstances in order to allow for such access or disclosure without violating any Law, Order or Contract or waiving attorney/client privilege or similar privilege.

"Environmental Law" shall mean any Law, Order or other requirement of Law (including Environmental Permits) that relates to (a) the protection of the environment (including natural resource restoration and natural resource damages) or of human health or safety (to the extent human health or safety relates to exposure to Hazardous Materials), or (b) the presence, Release, threatened Release, generation, recycling, disposal or treatment of Hazardous Materials, or the arrangement for any such activities.

"Environmental Liabilities" shall mean all Liabilities (including the costs of any Remedial Action) arising in connection with or in any way relating to the Business (as currently or formerly conducted), the Purchased Assets or any property currently or formerly owned, leased or operated in connection with the Business (as currently or formerly conducted) or the Purchased Assets (including the activities and operations conducted by anyone thereon, offsite disposal therefrom and Hazardous Materials migrating thereto or therefrom), that in each case arise under or relate to any Environmental Law, Environmental Permit or a Hazardous Material, including Liabilities arising from any Third Party claims for personal injury, death, or property damage caused by an actual or alleged release of, or exposure to, a Hazardous Material.

"EPA" shall mean the United States Environmental Protection Agency.

"Equity Commitment Letter" shall mean that certain equity commitment letter addressed to Purchaser, dated as of the date of the Initial Purchase Agreement, delivered by ArcLight Energy Partners Fund VI, L.P.

- "Escrow Account" shall mean the account created by the Escrow Agent in which the Good Faith Deposit Amount was deposited pursuant to the Initial Purchase Agreement.
- "Escrow Agent" shall mean Wells Fargo Bank, N.A. or, if Wells Fargo Bank, N.A. is unable or unwilling to act as the Escrow Agent, such other Person as is mutually agreed upon by Purchaser and Seller.
- "Escrow Agreement" shall mean that certain Escrow Agreement, dated as of October 8, 2015, by and among Purchaser, Seller and the Escrow Agent, as may thereafter be amended from time to time in accordance with the terms thereof.
- "Event" shall mean any action, omission, state of facts, condition, occurrence, result, circumstance, event, effect, development or change.
- "Excluded Furniture and Equipment" shall mean the Furniture and Equipment set forth on Section 1.1(a) of the Seller Disclosure Letter.
- "Excluded Intellectual Property" shall mean the rights to Intellectual Property listed on Section 1.1(b) of the Seller Disclosure Letter.
- "Excluded IT Assets" shall mean the rights to Information Technology listed on Section 1.1(c) of the Seller Disclosure Letter.
- "Excluded Permits" shall mean the Permits (and any pending applications) listed on Section 1.1(d) of the Seller Disclosure Letter.
- "Excluded Real Property" shall mean all real property owned by Seller, other than the Purchased Real Property, and including, for the avoidance of doubt, the Retained Refinery Assets and the Government Parcels.
- "Excluded Refinery Liabilities" shall mean (except to the extent covered in Section 2.3(n) or (o)) any Liabilities relating to or arising out of the Retained Refinery Assets, in each case relating to any act, omission, circumstance or other Event that occurs prior to or after the Closing, including to the extent exacerbated, triggered, increased, or their timing is accelerated by any act or omission of, or any delay caused by, Purchaser or any of its Affiliates, their respective successors or assigns or their respective Representatives.
- "Files and Records" shall mean all current and historical information, data, databases, reports, plans, schedules, correspondence, files, documents, books and other records and information, including (a) customer, vendor, supplier, contractor and service provider lists and records, (b) sales, pricing and performance data and records generated from completed or active transactions (including billing, payment and dispute histories, credit information and similar data), (c) business plans, financial statements, corporate, Tax, human resource, legal, regulatory and financial data and records, (d) operating and production records, quality control records, market research reports, technical documentation (drawings, specifications, process instructions, statistics, functional requirements, operating instructions, manual, etc.) and user documentation (installation guides, manuals, training material working paper, etc.), (e) employment and personnel records, (f) all of the plant diagrams and drawings, equipment

diagrams and drawings, equipment manuals, "as built" books, operating manuals, maintenance histories and records, civil engineering drawings, PFDs, PI&Ds and other such technical documents relating to the Above-Grade Refinery Assets and (g) regulatory filings, documents or other information relating to any Proceeding, in each case with respect to clauses (a) through (g), of Seller, whether in paper, electronic, digital or other form or medium, and that relate primarily to the Business, the Purchased Assets or Assumed Liabilities; provided, that Files and Records shall not include (i) any files, documents, books or other records to the extent relating to any existing or anticipated Tax litigations or disputes (to the extent that such litigation or disputes are, or are related to any, Excluded Liabilities hereunder), (ii) any information which, if transferred to Purchaser or its Affiliates, would reasonably be expected to violate any applicable Law or Order or the terms of any Contract and (iii) any items described in clauses (a) through (g) that relate primarily to the Retained Refinery Assets or any other Excluded Assets or any Excluded Liabilities.

"Final Order" shall mean an order or judgment of the Bankruptcy Court (or any other court of competent jurisdiction) entered by the clerk of the Bankruptcy Court (or such other court) on the docket in the Bankruptcy Case (or the docket of such other court), which is and remains in full force and effect, has not been modified, amended, reversed, vacated or stayed and as to which (i) the time to appeal, petition for certiorari, or move for a new trial, re-argument or rehearing has expired and as to which no appeal, petition for certiorari or motion for new trial, re-argument or rehearing shall then be pending or (ii) if an appeal, writ of certiorari new trial, reargument or rehearing thereof has been sought, such order or judgment of the Bankruptcy Court (or other court of competent jurisdiction) shall have been affirmed by the highest court to which such order was appealed, or certiorari shall have been denied, or a new trial, re-argument or rehearing shall have been denied or resulted in no modification of such order, and the time to take any further appeal, petition for certiorari or move for a new trial, re-argument or rehearing shall have expired, as a result of which such order shall have become final in accordance with Rule 8002 of the Federal Rules of Bankruptcy Procedure; provided, that the possibility that a motion under Rule 60 of the Federal Rules of Civil Procedure, or any analogous rule under the Federal Rules of Bankruptcy Procedure, may be filed relating to such order, shall not cause an order not to be a Final Order.

"Fourth Amendment Agreement" shall mean that certain Fourth Amendment Agreement, by and among the USVI Government, Seller, HOVIC and PDVSA VI, dated April 3, 2013, as ratified by the legislature of the Virgin Islands on November 4, 2013 and approved by the Governor of the Virgin Islands on November 4, 2013, as Act No. 7566 (such ratification including that certain letter, dated October 16, 2013, from George H.T. Dudley to the Governor of the Virgin Islands incorporated as part of Act No. 7566).

"Furniture and Equipment" shall mean all furniture, furnishings, equipment, vehicles, machinery, cranes, forklifts, tools, truck racks, fixtures, consumables and other tangible personal property owned by Seller, including artwork, desks, chairs, tables, miscellaneous office furnishings, copiers, scanners, printers, telephone lines and numbers, telecopy machines and other telecommunication equipment, cubicles and supplies.

"GAAP" shall mean generally accepted accounting principles of the United States of America consistently applied, as in effect from time to time.

"Good Faith Deposit Amount" shall mean \$19,000,000, which amount was deposited with the Escrow Agent in accordance with the terms of the Initial Purchase Agreement and held and released pursuant to the terms and subject to the conditions set forth in this Agreement and the Escrow Agreement.

"Governmental Entity" shall mean any multinational, United States or non-United States, federal, state, territory, provincial or local court (including, for the avoidance of doubt, the Bankruptcy Court), arbitral tribunal, administrative agency, legislature or commission or other governmental, quasi-governmental or regulatory agency or authority (including any bureau, division or department thereof) or any securities exchange.

"Hazardous Material" shall mean any waste or other substance that is listed, defined, designated, classified as, or otherwise determined to be, hazardous, extremely hazardous, toxic, radioactive, or a pollutant or a contaminant under or pursuant to any Law, Order or requirement of Law, including any admixture or solution thereof, and specifically including petroleum and all derivatives thereof or synthetic substitutes therefor and asbestos or asbestos-containing materials.

"Hovensa Operating Agreement" shall mean that certain Amended and Restated Limited Liability Company Agreement of Seller, dated as of October 30, 1998, and as thereafter amended from time to time, including those certain amendments dated as of February 10, 2000 and December 1, 2006.

"HOVIC" shall mean Hess Oil Virgin Islands Corp., a corporation organized under the Laws of the U.S. Virgin Islands.

"Improvements" shall mean any and all buildings, structures, fixtures or other improvements owned by or leased to Seller that are attached or affixed to the Business Real Property (including all construction and work-in-progress), including above ground and underground piping and storage tanks, canopies, signage, terminals, fixtures, pumps and appurtenances, control houses, the terminal office building, the administrative office building, the marine compound building, laboratory facilities, warehouses, boiler houses and waste water treatment facilities (and all equipment related thereto), the docks, equipment, personal property of a permanent nature and other similar facilities; provided, however, that the Retained Refinery Assets and Excluded Furniture and Equipment shall not be considered "Improvements".

"Indebtedness" of any Person shall mean (a) any obligations or indebtedness (i) for borrowed money or indebtedness issued or incurred in substitution or exchange for indebtedness for borrowed money, (ii) evidenced by any note, bond, debenture, mortgage or other debt instrument or debt security, (iii) for the deferred purchase price of property or service, (iv) for the reimbursement of any obligor on any letter of credit, banker's acceptance, guarantee or similar credit transaction (but only to the extent drawn upon or if there has been a demand made for reimbursement thereunder) and (v) under any commodity, swap, derivative, currency, interest rate, call, hedge or similar agreement, (b) any obligations of any kind referred to in the foregoing clause (a) above guaranteed or secured, directly or indirectly, in any manner by such Person (including by a Lien on the assets or properties of such Person whether or not such obligations are assumed) and (c) any accrued and unpaid principal, interest, premiums, penalties,

pre-payment penalties, fees, expenses, "make-whole" or similar payments, indemnities and breakage costs owing by such Person with respect to any indebtedness of a type described in clauses (a) or (b); provided, that Indebtedness of Seller shall not include accounts payable to trade creditors or accrued expenses, in each case, arising in the ordinary course of business of Seller (as currently conducted).

"Information Technology" shall mean all computers, computer systems, servers, workstations, databases, and software programs.

"Intellectual Property" shall mean (a) patents and patent applications, (b) trademarks, service marks, trade dress, and all registrations and applications for the same, (c) Internet domain names, (d) registered and unregistered copyrights and applications for the same, and (e) trade secrets.

"IRS" shall mean the United States Internal Revenue Service.

"Law" shall mean any statute, law, ordinance, ruling, policy, rule or regulation of any Governmental Entity and all judicial or administrative interpretations thereof and any common law doctrine.

"Leased Submerged Lands" shall mean those portions of real property identified in Exhibit H as "Terminal" or "Refinery" parcels that are leased by Seller subject to the Submerged Land Lease, the 1976 Contract and the 1998 Letter Agreement.

"Legal Proceeding" shall mean any judicial, administrative or arbitral actions, suits or legal proceedings (public or private) by or before a Governmental Entity.

"Liabilities" shall mean any and all Indebtedness, Taxes, losses, charges, debts, damages, obligations, payments, costs and expenses, bonds, indemnities, liabilities and obligations of any nature, including any unknown, undisclosed, unmatured, unaccrued, unasserted, contingent, indirect, conditional, implied, vicarious, derivative, joint, several or secondary liability, regardless of whether such claim, debt, obligation, duty or liability would be required to be disclosed on a balance sheet prepared in accordance with GAAP and regardless of whether such claim, debt, obligation, duty or liability is immediately due and payable.

"Liens" shall mean any liens (as defined in Section 101(37) of the Bankruptcy Code), debts (as defined in Section 101(12) of the Bankruptcy Code), security interests, Claims, easements, mortgages, charges, indentures, deeds of trust, rights of way, encroachments, or any other encumbrances and other restrictions or limitations on ownership or use of real or personal property or irregularities in title thereto.

"Loss" or "Losses" shall mean, without duplication, any and all Liabilities, judgments, awards, losses, costs or damages, including reasonable fees and expenses of attorneys, accountants and other professional advisors.

"LPG Flare" shall mean the liquid petroleum gas flaring facility and related interconnection equipment currently used in the operation of the Business.

"Material Adverse Effect" shall mean any Event that, individually or in the aggregate with any other related Events, (a) has had or would reasonably be expected to have a material adverse effect on the business, condition (financial or otherwise), properties, assets or results of operations of the Business, the Purchased Assets or the Assumed Liabilities, in each case, taken as a whole or (b) prevents or materially delays beyond the End Date, or would reasonably be expected to prevent or materially delay beyond the End Date, the consummation of the transaction contemplated by this Agreement or the Transaction Documents or the performance of any of Seller's material obligations under this Agreement or the Transaction Documents; provided, that no Event (by itself or when aggregated with other Events) to the extent resulting from any of the following, shall be considered in determining whether there has been or would reasonably be expected to be a Material Adverse Effect:

- (i) changes in general economic, business or political conditions;
- conditions or changes in the securities markets, credit markets, (ii) currency markets or other financial markets in the United States or any other country or region in the world, including (x) interest rates in the United States or any other country or region in the world and exchange rates for the currencies of any countries and (y) any suspension of trading in securities (whether equity, debt, derivative or hybrid securities) on any securities exchange or over-the counter market operating in the United States or any other country or region in the world;
- changes in any applicable Laws or interpretations thereof by any Governmental Entity or GAAP;
- conditions affecting generally the hydrocarbon, terminal or refining industry (including feedstock pricing, refining, marketing, transportation, terminalling, trading costs and margins, and crude oil refined products and other commodity prices) or any other industries or markets in which the Business, the Purchased Assets and the Assumed Liabilities are owned or operated;
- other than for purposes of any representation or warranty set forth in Section 4.2(b) or Section 4.3, the announcement of this Agreement or the consummation of the transactions contemplated by this Agreement;
 - (vi) the commencement or continuation of the Bankruptcy Case;
- any action required to be taken by the Bankruptcy Court pursuant to the Sale Motion, the Sale Order, or the Bid Procedures Order;
- (viii) general political conditions in the United States, the U.S. Virgin Islands or any other country or region in the world or any natural or man-made disaster or any acts of terrorism, sabotage, military action or war (whether or not declared) or any escalation or worsening thereof, regardless of when occurring or commenced (and excluding, for the avoidance of doubt, the effect of the failure of any Governmental Entity to approve any transactions contemplated by this Agreement or the Transaction Documents);

- (ix) any failure by Seller or any of its Affiliates or the Business (or any portion thereof) to meet any estimates or expectations of revenue, earnings or other financial performance or results of operations for any period, or any failure to meet internal or published projections, budgets, plans or forecasts of revenues, earnings, cash flows or other financial performance or results of operations for any period; provided, that the exception in this clause shall not prevent the underlying cause or causes of such failure from being taken into consideration in determining whether there has been or would reasonably be expected to be a Material Adverse Effect; or
- (x) any action taken at the written request of, or with the express prior written consent of, Purchaser;

provided, further, that any Event in clauses (i), (ii), (iii), (iv) and (viii) may be taken into account when determining whether there has been or would reasonably be expected to be a Material Adverse Effect to the extent that such Event has a disproportionate effect on the Business, the Purchased Assets and the Assumed Liabilities taken as a whole, compared to other hydrocarbon terminalling and storage businesses in the United States or Caribbean region. A "Material Adverse Effect" shall not be measured based on any forward-looking statements, projections or forecasts of Seller or any other Person.

"Notes" shall mean the 2012 Notes and the 2015 Notes.

"NRD Settlement Agreement" shall mean that certain Settlement and Release Agreement entered into on May 29, 2014 by Seller, Alicia V. Barnes in her capacity as Trustee for Natural Resources of the Territory of the United States Virgin Islands, the USVI Government in its *parens patriae* and public trustee capacity, on behalf of the public and its quasi-sovereign interests, and HOVIC.

"Operational Terminals Business" shall mean the ownership and operation of a hydrocarbon storage and terminal facility consistent in all material respects with the manner in which the Business was operated by Seller when the Business was fully operational, assuming such ownership and operation are in compliance with all Laws, Permits, Orders and Contracts.

"Option Agreement" shall mean an option agreement by and among Purchaser and Seller, pursuant to which Purchaser will have the right to acquire some or all of the Option Refinery Property for a purchase price of \$1 per acre, exercisable until the date that is the earlier of (i) ten (10) years from the Closing Date and (ii) three (3) years after the removal of the Above-Grade Refinery Assets from the Option Refinery Property and which is otherwise in form and substance reasonably satisfactory to each of Seller and Purchaser.

"Option Refinery Property" shall mean the real property owned by Seller and described in Section 1.1(e) of the Seller Disclosure Letter (and including, for the avoidance of doubt, the Option Refinery Parcels), together with all easements, appurtenances, rights and other hereditaments appurtenant to such real property.

"Order" shall mean any judgment, order, injunction, decree, writ, permit or license issued or entered by or with any Governmental Entity or any arbitrator, whether

preliminary, interlocutory or final, including any order entered by the Bankruptcy Court in the Bankruptcy Case (including the Sale Order).

"OSRO" shall mean Oil Spill Response Organization.

"Parts Inventory" shall mean the inventories of supplies, maintenance and capital spares, joints, valves, parts and tools owned by Seller as of the Closing that are (x) stored in the warehouses located on the Business Real Property or in the Above-Grade Refinery Assets or (y) used primarily in connection with the Business; provided, that the Parts Inventory shall not include any Retained Refinery Assets.

"PDVSA VI" shall mean PDVSA V.I., Inc., a corporation organized under the Laws of the U.S. Virgin Islands.

"Pension Plan" shall mean that certain HOVENSA L.L.C. Employees' Pension Plan.

"Permitted Liens" shall mean (a) statutory Liens or other Liens arising in the ordinary course of business of Seller (including by operation of law) securing payments not yet due, including mechanics', carriers', workmen's, repairmen's, materialmens' and maritime Liens, (b) Liens for Taxes not yet due and payable or for Taxes that may thereafter be paid without penalty or which are being contested in good faith and by appropriate Proceedings and are set forth on Section 1.1(f) of the Seller Disclosure Letter, (c) Liens set forth in Section 1.1(f) of the Seller Disclosure Letter, (d) Liens affecting the Business Real Property with respect to (i) zoning, building code or planning restrictions or regulations, servitudes, permits, licenses, surface leases, ground leases to utilities, municipal agreements, railway siding agreements and other similar rights, easements for streets, alleys, highways, telephone lines, gas pipelines, power lines and railways, and other easements and rights of way of public record on, over, or in respect of any such real property and (ii) encroachments and other matters that would be shown in an accurate survey or physical inspection of such real property that, in each case of (i) and (ii) of this clause (d) above, do not, individually or in the aggregate, materially interfere with the use, operation or occupancy of such real property as an Operational Terminals Business, (e) Liens arising pursuant to any Environmental Permit, (f) any Liens arising under ERISA or the Code with respect to the Pension Plan and (g) any Lien that will be extinguished at or prior to Closing.

"Permitted Tax Liens" shall mean those Liens, and only those Liens, described in clause (b) of the definition of Permitted Liens.

"Person" shall mean and include an individual, a partnership, a limited partnership, a limited liability partnership, a joint venture, a corporation, a limited liability company, an association, a trust, an unincorporated organization, a group and a Governmental Entity.

"Petition" means the voluntary petition under Chapter 11 of the Bankruptcy Code filed by Seller with the Bankruptcy Court.

"Petition Date" shall mean September 15, 2015.

- "Pinnacle Contract" shall mean that certain Term Services Agreement, dated as of May 1, 2012, by and between Seller and Pinnacle Services L.L.C., as amended from time to time in accordance with its terms.
- "Pinnacle Cure Amount" shall mean the Cure Amount with respect to the Pinnacle Contract.
- "Plant Closing Act" shall mean the Virgin Islands Code Title 24, § 471- 478 (1957).
- "Post-Closing Litigation Liabilities" shall mean all Liabilities arising from a Proceeding first commenced after the Closing by a Third Party against Purchaser, Seller or any of their respective Affiliates, in respect of the Business, the Purchased Assets or the Assumed Liabilities that is based on any Event first existing, arising or occurring following the Closing; provided, that the Parties agree and acknowledge that Post-Closing Litigation Liabilities shall not include any Environmental Liabilities, which are separately addressed in this Agreement.
- "Post-Closing Period" shall mean all taxable years or other taxable periods that begin after the Closing Date and, with respect to any taxable year or other taxable period beginning on or before and ending after the Closing Date, the portion of such taxable year or period beginning on the day after the Closing Date.
- "Pre-Closing Litigation Liabilities" shall mean all Liabilities arising from a Proceeding first commenced prior to or following the Closing by a Third Party against Purchaser, Seller or any of their respective Affiliates in respect of the Business, the Purchased Assets or the Assumed Liabilities that is based on any Event first existing, arising or occurring before the Closing; provided, that the Parties agree and acknowledge that Pre-Closing Litigation Liabilities shall not include any Environmental Liabilities, which are separately addressed in this Agreement.
- "Pre-Closing Period" shall mean all taxable years or other taxable periods that end on or before the Closing Date and, with respect to any taxable year or other taxable period beginning on or before and ending after the Closing Date, the portion of such taxable year or period ending on the Closing Date.
- "Proceeding" shall mean any claim, demand, action, arbitration, audit, hearing, investigation, litigation, suit or other proceeding (whether civil, criminal, administrative, investigative, or informal) commenced, brought, conducted, or heard by or before, or otherwise involving, any Third Party (including any Governmental Entity).
- "Property Taxes" shall mean all real property Taxes, personal property Taxes and other similar ad valorem Taxes, in each case, that are imposed with respect to the Purchased Assets.
- "Protocol of Delivery and Acceptance" shall mean, with respect to each Tug Boat, the applicable protocol of delivery and acceptance for such Tug Boat, substantially in the form attached hereto as Exhibit D.

"Purchase Price" shall mean \$190,000,000.

"Purchased Furniture and Equipment" shall mean all Furniture and Equipment owned by Seller that is located at the Seller Facilities or the Above-Grade Refinery Assets or that is otherwise primarily used or held for use in the Business (assuming that the Business includes an Operational Terminals Business) or in the ownership or operation of the Above-Grade Refinery Assets; provided, that, for the avoidance of doubt, the Purchased Furniture and Equipment shall not include the Excluded Furniture and Equipment or any other Excluded Assets.

"Purchaser Expense Reimbursement" shall mean the sum of the aggregate amount of the Purchaser's reasonable documented out-of-pocket costs and expenses (including expenses of outside counsel, accountants and financial advisors, which shall be based on summary invoices, redacted to preserve privileged or confidential information, and which shall not be required to comply with applicable United States Trustee guidelines) incurred by Purchaser in connection with or related to Purchaser's evaluation, consideration, analysis, negotiation, and documentation of the transactions contemplated by this Agreement, up to a maximum amount of \$1,900,000.

"RCRA Permit" shall mean the Resource Conservation and Recovery Act Part B Permit No. VID980536080 and any document that replaces such Permit.

"Real Property" shall mean the Purchased Real Property and the Excluded Real Property.

"Release" shall mean the disposing, discharging, injecting, spilling, leaking, pumping, leaching, dumping, emitting, escaping or emptying into or upon any air, soil, sediment, subsurface strata, surface water or groundwater.

"Remedial Action" shall mean any action to investigate, abate, clean up, remove or remediate (or words of similar import), or conduct remedial or corrective actions, including sampling and/or monitoring activities, with respect to Hazardous Materials in the environment.

"Representatives" of any Person shall mean such Person's directors, managers, officers, employees, agents, attorneys, consultants, advisors, financing sources or other Persons acting on behalf of such Person.

"Restructuring Transaction" shall mean, in each case, which does not include the Purchase and Tug Boat Sale pursuant to this Agreement (i) any recapitalization transaction, plan of reorganization, liquidation, or sale, including any such transaction by way of a credit bid or by any creditor of Seller, involving, whether in whole or in part, Seller or all or any material portion of the Purchased Assets, or (ii) any merger, consolidation, share exchange, business combination or similar transaction involving, whether in whole or in part, Seller or all or any material portion of the Purchased Assets, in each case whether in one transaction or a series of transactions.

"Retained Refinery Assets" shall mean the Option Refinery Property, to the extent such Option Refinery Property has not been conveyed pursuant to the Option Agreement (and excluding, for the avoidance of doubt, the Above-Grade Refinery Assets).

"Returns" shall mean all returns, statements, forms, reports and other similar documentation relating to Taxes, including any schedules, exhibits and other attachments thereto and any amendments thereof.

"Sale Hearing" shall mean the hearing at which the Bankruptcy Court considers approval of the Sale Order pursuant to Sections 105, 363 and 365 of the Bankruptcy Code.

"Sale Motion" shall mean Debtor's Motion for Entry of Orders (a)(i) Establishing Bidding Procedures Relating to the Sale of the Debtor's Assets, Including Approving Break-Up Fee and Expense Reimbursement, (ii) Establishing Procedures Relating to the Assumption and Assignment of Certain Executory Contracts and Unexpired Leases, Including Notice of Proposed Cure Amounts, (iii) Approving Form and Manner of Notice Relating Thereto, and (iv) Scheduling a Hearing to Consider the Proposed Sale; (b)(i) Approving the Sale of the Debtor's Assets Free and Clear of All Liens, Claims, Encumbrances, and Interests, and (ii) Authorizing the Assumption and Assignment of Certain Executory Contracts and Unexpired Leases; and (c) Granting Related Relief filed by the Seller in the Bankruptcy Court on September 15, 2015.

"Sale Order" shall mean an Order of the Bankruptcy Court approving this Agreement, containing the findings and conclusions set forth in Section 7.15, and authorizing and directing Seller to consummate the transactions contemplated by this Agreement (including, but not limited to, the assumption and assignment of the Business Contracts and Real Property Leases) under sections 105, 363 and 365 of the Bankruptcy Code, in form and substance acceptable to Seller and Purchaser in their respective sole discretion.

"Sanctions" shall mean any sanctions or embargos administered or enforced by the U.S. Government, (including, the OFAC Regulations), the United Nations Security Council, the European Union, Her Majesty's Treasury, or other relevant Governmental Entity that is a sanctions authority.

"Seller Facilities" shall mean the storage, terminalling and refining facilities and related equipment owned by Seller located on the Business Real Property and the Option Refinery Property, including all Improvements related thereto; provided, that, for the avoidance of doubt, the Seller Facilities shall not include any Excluded Assets.

"Seller Letters of Credit" shall mean (a) that certain Irrevocable Letter of Credit No. 123146-793, issued as of April 5, 2012 by Intesa Sanpaolo S.p.A. on behalf of Hess Corporation for the benefit of the EPA, (b) that certain Irrevocable Letter of Credit No. 123147-793, issued as of April 5, 2012 by Intesa Sanpaolo S.p.A. on behalf of Hess Corporation for the benefit of the EPA, (c) that certain Irrevocable Standby Letter of Credit No. CNYI 5051/12, issued as of April 4, 2012 by Banco Espirito Santo on behalf of Seller for the benefit of the EPA, and (d) that certain Irrevocable Standby Letter of Credit No. CNYI 5050/12, issued as of April 4,

2012 by Banco Espirito Santo on behalf of Seller for the benefit of the EPA, and any document that replaces such Letters of Credit.

"Shared Services Agreement" shall mean a shared services agreement by and among Purchaser and Seller, in form and substance satisfactory to Seller and Purchaser in their sole discretion.

"Submerged Land Lease" shall mean that certain Lease, dated as of October 16, 1976, by and between the USVI Government and Seller (as assignee of HOVIC).

"Subsidiary", with respect to any Person, shall mean (a) any corporation more than fifty percent (50%) of the stock of any class or classes of which having by the terms thereof ordinary voting power to elect a majority of the directors of such corporation (irrespective of whether or not at the time stock of any class or classes of such corporation shall have or might have voting power by reason of the happening of any contingency) is owned by such Person directly or indirectly through one or more subsidiaries of such Person and (b) any partnership, association, joint venture, limited liability company or other entity in which such Person directly or indirectly through one or more subsidiaries of such Person has more than a fifty percent (50%) equity interest.

"Taxes" shall mean all taxes, assessments, charges, duties, fees, levies or other governmental charges imposed by a taxing authority, including all United States federal, state, territory (including the U.S. Virgin Islands), local, foreign and other income, franchise, profits, gross receipts, capital gains, capital stock, transfer, sales, use, value added, occupation, property, excise, severance, windfall profits, stamp, license, payroll, social security, withholding and other taxes, assessments, charges, duties, fees, levies or other governmental charges imposed by a taxing authority of any kind whatsoever (whether payable directly or by withholding and whether or not requiring the filing of a Return), all estimated taxes, deficiency assessments, additions to tax, penalties and interest, and including Liability for any of the foregoing items pursuant to applicable Law, as a transferee, as a successor, or pursuant to a Contract.

"Terminal Assets" shall mean the Purchased Assets, other than the Above-Grade Refinery Assets.

"Third Party" shall mean any Person other than the Parties and their respective Affiliates (including, for the avoidance of doubt, any Governmental Entity).

"Transaction Documents" shall mean the Access Easement, the Escrow Agreement, the Shared Services Agreement, the Ground Leases, the Cross-Easement Agreements and the Option Agreement.

"Tug Boat Bill of Sale and Acceptance" shall mean, with respect to each Tug Boat, the applicable Bill of Sale and Acceptance of sale for such Tug Boat, substantially in the form attached hereto as Exhibit E, pursuant to which HOVIC will transfer to Purchaser at the Closing such Tug Boat.

"Tug Boats" shall mean the following six (6) tug boats owned by HOVIC: (1) M/V Turquoise Bay, (2) M/V Teague Bay, (3) M/V Canegarden Bay, (4) M/V Manchenil Bay, (5) M/V Limetree Bay and (6) M/V HOVIC 1.

"USCG" shall mean the United States Coast Guard.

"USVI Concession Agreement" shall mean that certain concession agreement substantially in the form attached hereto as Exhibit F.

"USVI Government" shall mean the Government of the United States Virgin Islands, including any United States Virgin Islands Governmental Entity.

"USVI Government Concession Fee" shall mean \$100,000,000. The USVI Government Concession Fee reflects the amount necessary to satisfy all monetary obligations of Seller to the USVI Government in respect of the Purchased Assets, including the full and final satisfaction and settlement of the secured claim in the amount of approximately \$40,000,000, as certified by Seller and the USVI Government at least ten (10) days prior to the Closing.

Additional Defined Terms. In addition to the terms defined in Section 1.1, additional defined terms used herein shall have the respective meanings assigned thereto in the Sections indicated below.

Defined Term	Section	
1976 Contract Permits	Seller Disclosure	
	Letter	
Additional Requirements	7.24	
Agreement	Preamble	
Allocation	3.3(b)	
Assumed Liabilities	2.3	
Bankruptcy Case	Recitals	
Bankruptcy Court	Recitals	
Bankruptcy Rules	4.3	
Business	Recitals	
Business Confidential Information	7.22	
Business Contracts	2.1(a)(i)	
Business Permits	2.1(a)(vii)	
Closing	3.5(a)	
Closing Cross-Easement Agreements	7.28	
Closing Date	3.5(a)	
Closing Payment	3.1	
Deferred Asset	2.6(a)	
Deferred Terminal Parcels	7.28	
Designation Deadline	2.5(a)(iii)	
Efforts Condition	7.4(a)	
Eliminated Agreement	2.5(a)(iii)	
Employee Benefit Plans	4.9	
End Date	9.1(b)(ii)	

Case 1:15-bk-10003-MFW Doc 394-1 Filed 12/01/15 Entered 12/01/15 12:35:34 Desc Exhibit Page 25 of 49

Environmental Permits	4.12(c)
Equity Financing	6.4
Equity Financing Source	6.4
ERISA	4.9
Excess Power	7.24
Excluded Assets	2.2
Excluded Liabilities	2.4
FIRPTA	3.4
Government Parcels	7.28
Ground Lease	7.28
Hovensa Consultant	4.11
HOVIC	Preamble
HOVIC Disclosure Letter	Article V
HOVIC Update Schedule	7.7(a)
Initial Purchase Agreement	Recitals
Knowledge of Purchaser	1.5
Knowledge of Seller	1.5
Limited Consent Decree Modification	7.4(c)
Material Contracts	4.6(b)
Other Required Approvals	8.2(d)
Option Refinery Parcels	7.28
Parties	Preamble
Permits	4.14
Permitting Process	7.25
Pinnacle Employee	7.9
Purchase	Recitals
Purchased Assets	2.1(a)
Purchased Real Property	2.1(a)(ii)
Purchaser	Preamble
Purchaser Borne Withholding Taxes	3.4
Purchaser's Required Permits	7.18(d)
Real Property Leases	2.1(a)(iv)
Related Parties	11.14
Released Claims	10.2
Releasee	10.2
Releasor	10.2
Required Governmental Approvals	8.1(e)
Seller	Preamble
Seller Access Indemnitees	7.1(b)
Seller Disclosure Letter	Article IV
Seller Update Schedule	7.7(a)
Seller's Marks	7.23
Subdivision Parcels	7.28
Tax Purchase Price	3.3(a)
Transfer Taxes	7.10(a)
Trust Funds	2.2(q)

Case 1:15-bk-10003-MFW Doc 394-1 Filed 12/01/15 Entered 12/01/15 12:35:34 Desc Exhibit Page 26 of 49

Tug Boat Permits	7.25
Tug Boat Sale	Recitals
Update Schedule	7.7(a)
WARN	7.9
Wind-Up Cost Cap	7.33
Wind-Up Costs	7.33

1.3 <u>Construction</u>. In this Agreement, unless the context otherwise requires:

- (a) references to "writing" or comparable expressions include a reference to electronic mail, provided that, where applicable, the sender complies with the provisions of Section 11.3;
- (b) the phrases "delivered" or "made available" shall mean that the information referred to has been physically or electronically delivered to the relevant parties (including, in the case of "made available" to Purchaser, material that has been posted and thereby made available to Purchaser through the on-line "virtual data room" established by Seller);
- (c) words expressed in the singular number shall include the plural and vice versa; words expressed in the masculine shall include the feminine and neuter gender and vice versa;
- (d) references to Articles, Sections, Sections of the Seller Disclosure Letter, Sections of the HOVIC Disclosure Letter, Exhibits, the Preamble and Recitals are references to articles, sections, exhibits, the preamble and recitals of this Agreement and the disclosure letters delivered with respect to this Agreement, and the descriptive headings of the several Articles and Sections of this Agreement, the Seller Disclosure Letter and the HOVIC Disclosure Letter (as applicable) are inserted for convenience only, do not constitute a part of this Agreement and shall not affect in any way the meaning or interpretation of this Agreement;
 - (e) references to "day" or "days" are to calendar days;
- (f) the words "hereof", "herein", "hereto" and "hereunder", and words of similar import, shall refer to this Agreement as a whole and not to any provision of this Agreement;
- (g) this "Agreement" or any other Contract or document shall be construed as a reference to this Agreement or, as the case may be, such other Contract or document as the same may have been, or may from time to time be, amended, varied, novated or supplemented and shall include each and every exhibit, attachment, schedule, addendum, appendix, statement of work, change order, and any other similar instrument relating to such Contract or document (as amended);
- (h) "include", "includes", and "including" are deemed to be followed by "without limitation" whether or not they are in fact followed by such words or words of similar import;

- (i) references to "Dollars", "dollars" or "\$", without more are to the lawful currency of United States of America;
- (j) references to "commercially reasonable efforts" shall not require any Party to repay any Indebtedness, amend any Contract to increase the amount payable thereunder or otherwise to be materially more burdensome to such Party, commence any litigation, settle or compromise any matter, offer or grant any accommodation (financial or otherwise, including with respect to the Notes or any accounts payable) to any Third Party, pay any amount or bear any other incremental economic burden and, without limiting the foregoing, for purposes of Section 7.11(a), references to "commercially reasonable efforts" shall not require any Party to agree or consent to any modifications to the form of USVI Concession Agreement attached as Exhibit F;
- (k) with respect to the Business (including the Purchased Assets and Assumed Liabilities), references to the "ordinary course of business" or its business in the "ordinary course" shall mean (i) with respect to the period from and after February 5, 2015, the ordinary course of business of Seller as currently conducted, (ii) with respect to the period from February 21, 2012 until February 5, 2015, the ordinary course of business of Seller solely as the owner and operator of a hydrocarbon storage terminal (taking into account changes in the business of Seller in preparation for the shutdown of the terminalling and storage business) and (iii) with respect to all periods prior to February 21, 2012, the ordinary course of business of Seller as the owner and operator of a petroleum refining business (taking into account changes in the business of Seller in preparation for the shutdown of its refinery operations); and
- (l) references to any Party shall include such Party's successors and permitted assigns.
- 1.4 <u>Exhibits and the Disclosure Letters</u>. The Exhibits, the Seller Disclosure Letter and the HOVIC Disclosure Letter are incorporated into and form an integral part of this Agreement.
- 1.5 <u>Knowledge</u>. When any representation, warranty, covenant or agreement contained in this Agreement is expressly qualified by reference to the "**Knowledge of Seller**" or words of similar import with respect to Seller, it shall mean the actual knowledge of the individuals set forth in <u>Section 1.5</u> of the Seller Disclosure Letter with respect to Seller, without inquiry or investigation. When any representation, warranty, covenant or agreement contained in this Agreement is expressly qualified by reference to the "**Knowledge of Purchaser**" or words of similar import, it shall mean the actual knowledge of Evan Schwartz, Jake Erhard and Christine Miller, without inquiry or investigation.

ARTICLE II

SALE OF ASSETS

2.1 Sale of Purchased Assets and Tug Boats.

(a) On the terms and subject to the conditions of this Agreement (including as set forth in Section 2.5(b)) and the Sale Order, Purchaser agrees to purchase from Seller, and

Seller agrees to sell, convey, transfer, assign and deliver to Purchaser, at the Closing, all the right, title and interest of Seller in and to the Purchased Assets, free and clear of any Liens, Claims and Liabilities of any kind whatsoever except Permitted Liens. The "Purchased Assets" shall mean all right, title and interest of Seller in the following assets (other than any such assets excluded pursuant to Section 2.2)

- each Contract listed or described in Section 2.1(a)(i) of the Seller Disclosure Letter and the Pinnacle Contract (collectively, the "Business Contracts");
- the real property described in Section 2.1(a)(ii) of the Seller (ii) Disclosure Letter, together with all easements, appurtenances, rights and other hereditaments appurtenant to such real property (collectively, the "Purchased Real Property"); provided, that any Option Refinery Property transferred to Purchaser pursuant to the Option Agreement shall be considered Purchased Real Property (and therefore shall be Business Real Property) for purposes of this Agreement;
 - (iii) all Improvements;
- each real property lease and sublease listed or described in Section (iv) 2.1(a)(iv) of the Seller Disclosure Letter ("Real Property Leases");
 - (v) all Parts Inventory;
- the Files and Records, whether in hard copy or electronic format, except that, with respect to Files and Records that Seller or any of its Affiliates is required by Law or Order to retain, the Purchased Assets shall include copies of such Files and Records to the extent permitted by any applicable Law or Order (and, with respect to any personnel records which Seller is required to retain, copies of such personnel records to the extent permitted by any applicable Law or Order); provided, that Seller shall be permitted to retain a copy of any such Files and Records that Seller requires in connection with any of the Excluded Assets;
- all Permits (and any pending applications), other than the (vii) Excluded Permits, that are owned, utilized, held or maintained by or licensed to Seller primarily in connection with Seller's ownership or operation of the Business, the other Purchased Assets or the Assumed Liabilities or which are otherwise required for Seller's operation of the Business or the other Purchased Assets (assuming, in each case, that the Business includes an Operational Terminals Business), including those Permits described in Section 2.1(a)(vii) of the Seller Disclosure Letter (which shall not include the RCRA Permit) (the "Business Permits");
 - (viii) the Business Intellectual Property;
 - the Business IT Assets; (ix)
- all rights, claims, defenses, causes of action (including Avoidance Actions) and rights of offset or counterclaim against a Third Party relating to or arising from the Business, the Purchased Assets or the Assumed Liabilities (including rights

under manufacturer and vendor warranties, indemnities and guarantees), except to the extent related to any Excluded Assets or Excluded Liabilities; provided, that Purchaser shall not prosecute, or commence any litigation with respect to, the Avoidance Actions;

- the Purchased Furniture and Equipment; (xi)
- (xii) accounts receivable and pre-paid assets, solely to the extent related to an Assumed Liability;
- (xiii) all tank bottoms, heels and line fill which are located at the Seller Facilities:
 - (xiv) the LPG Flare;
 - (xv) the Above-Grade Refinery Assets; and
- (xvi) with respect to the Tug Boats, the Tug Boat spare parts owned by Seller and stored (x) in the warehouses and the marine compound located on the Business Real Property or in the Above-Grade Refinery Assets and (y) on the Tug Boats.
- On the terms and subject to the conditions of this Agreement, Purchaser agrees to purchase from HOVIC, and HOVIC agrees to sell, convey, transfer, assign and deliver to Purchaser, at the Closing, all its right, title and interest in and to the Tug Boats, together with, to the extent transferable, the related assets described in Section 2.1(b) of the HOVIC Disclosure Letter.
- 2.2 Excluded Assets. Notwithstanding anything herein to the contrary, the Purchased Assets shall not include any assets, rights or privileges of Seller other than the Purchased Assets and specifically shall not include Seller's right, title or interest in any of the following assets, whether owned by, held by or relating to Seller (collectively, the "Excluded Assets"):
 - (a) all cash, certificates of deposit or other cash equivalents;
- (b) any real property other than (i) Purchased Real Property and (ii) real property subject to the Real Property Leases;
- all personal property, equipment and inventory not included in the Purchased Assets;
- all rights, claims, defenses, causes of action and rights of offset or (d) counterclaim (at any time or in any manner arising or existing, whether choate or inchoate, known or unknown, contingent or non-contingent) of Seller against Third Parties (i) to the extent relating to any of the Excluded Assets or Excluded Liabilities or (ii) not included in the Purchased Assets:
- all Avoidance Actions related solely to any Excluded Assets or Excluded (e) Liabilities;

Case 1:15-bk-10003-MFW Doc 394-1 Filed 12/01/15 Entered 12/01/15 12:35:34 Desc Exhibit Page 30 of 49

- (f) all rights under any Contract to which Seller is a party that is not a Business Contract or a Real Property Lease;
- (g) general books of account and books of original entry that comprise Seller's permanent Tax records, corporate minute books, stock books and related organizational documents and the books and records that Seller is required to retain pursuant to any Law or Order and the books and records solely related to the Excluded Assets or Excluded Liabilities;
- (h) personnel records that Seller is required by Law or Order to retain (provided that the copies of any such personnel records shall be Purchased Assets to the extent permitted by any applicable Law or Order pursuant to Section 2.1(a)(vi)) and personnel records of employees, former employees or consultants of Seller who do not, as of the Closing, become employees of Purchaser;
- (i) all insurance recoveries under Seller's or its Affiliates' insurance policies with respect to the Purchased Assets or the Business and any rights to assert claims with respect to any such insurance recoveries;
- (j) all claims for refund or credit of (i) income Taxes of Seller and its Affiliates for any taxable period and (ii) non-income Taxes of Seller and its Affiliates with respect to a Pre-Closing Period;
 - (k) the Excluded Intellectual Property;
 - (1) the Excluded IT Assets;
- (m) Seller's or any of its Affiliates' rights under this Agreement or any Transaction Document or relating to the Excluded Assets or the Excluded Liabilities;
- (n) all deposits (including security deposits, rent, electricity, telephone or otherwise and retainers held by attorneys, accountants, financial advisors and other professional advisors retained by Seller or by any creditors' committee in the Bankruptcy Case) and other prepaid charges of Seller, in each case other than with respect to any Purchased Assets;
- (o) any assets of Seller in a directors and officers liability insurance policy, executive or incentive compensation, bonus, deferred compensation, pension, retiree medical, stock option or other stock purchase plan or other Employee Benefit Plan;
 - (p) the RCRA Permit and any related financial assurance;
- (q) any rights or assets related to trusts for groundwater remediation or land farms (the "Trust Funds");
- (r) all toy trucks and related inventory which are located at the Seller Facilities;
 - (s) all Excluded Furniture and Equipment;

- (t) all Retained Refinery Assets;
- (u) all accounts receivable and pre-paid assets of Seller, except to the extent related to a Purchased Asset or an Assumed Liability;
 - (v) the Excluded Permits; and
- (w) all attorney client privilege and attorney work product protection of Seller or associated with the Business (as currently or formerly conducted) as a result of legal counsel representing Seller or the Business (as currently or formerly conducted), and all Files and Records related thereto subject to such attorney client privilege or work product protection; provided that any privilege or work product that exclusively relates to the Purchased Assets or the Assumed Liabilities or that is necessary to operate the Business (other than such privileges, protections and related Files and Records related to or in connection with the transactions contemplated by this Agreement or any Transaction Document) shall not be deemed to be an Excluded Asset whether or not in connection with the transactions contemplated by this agreement or any Transaction Document.
- 2.3 <u>Assumption of Liabilities</u>. On the terms and subject to the conditions of this Agreement, including <u>Section 2.5(b)</u>, and except for the Excluded Liabilities, Purchaser agrees, effective as of the Closing (except as otherwise provided), to assume and shall agree to pay, perform and discharge when due, the following Liabilities (collectively, the "Assumed Liabilities") of Seller:
- (a) all Liabilities (including for the avoidance of doubt, all Environmental Liabilities), other than Excluded Refinery Liabilities, relating to or arising out of the Purchased Assets or the Business arising out of or relating to any act, omission, circumstance or other Event occurring after the Closing, provided, however, that, except with respect to Cure Amounts assumed pursuant to Section 2.3(c), Purchaser shall not assume or agree to pay, discharge or perform any Liabilities of Seller under or with respect to any Business Contracts and Real Property Leases, including Liabilities arising out of any breach, misfeasance or under any other theory to the extent relating to Seller's conduct prior to the Closing;
 - (b) all Post-Closing Litigation Liabilities;
- (c) all Cure Amounts with respect to the Business Contracts and the Real Property Leases (other than the Pinnacle Cure Amount);
 - (d) all valid reclamation claims related to the Business;
- (e) any and all costs and expenses necessary in connection with providing "adequate assurance of future performance" with respect to the Business Contracts and Real Property Leases (as contemplated by section 365 of the Bankruptcy Code) for the period commencing on or after the Closing;
- (f) all Environmental Liabilities, other than Excluded Refinery Liabilities, arising out of or relating to any act, omission or circumstance that occurred prior to the Closing, including for the avoidance of doubt, the presence of Hazardous Materials arising from Seller's

former refinery operations and located at the Purchased Assets, solely to the extent such Liabilities are exacerbated, triggered, increased, or have their timing accelerated by any act or omission of, or any delay caused by, Purchaser or any of its Affiliates, their respective successors or assigns or their respective Representatives; provided, that, all Environmental Liabilities arising out of or relating to Events or conditions occurring or existing prior to Closing shall not be Assumed Liabilities solely by virtue of Purchaser discovering, identifying or quantifying such Environmental Liabilities through sampling of environmental media if such sampling is (i) required by Environmental Law or by a Governmental Entity, (ii) required to be conducted in response to a Third Party Claim alleging that Hazardous Materials have migrated offsite from the Business Real Property or (iii) undertaken in connection with ordinary course construction, remodeling, demolition, operation or maintenance, assuming that the Business includes an Operational Terminals Business, or undertaken in connection with the dismantling, demolition or removal of the Above-Grade Refinery Assets:

- all Liabilities related to the Business Contracts and Real Property Leases arising out of or relating to any act, omission, circumstance or other Event occurring on or after the Closing Date (including the ordinary course obligations arising under the Business Contracts and Real Property Leases in accordance with their terms);
- all accounts and notes payable to Seller's third-party trade creditors that relate to the Purchased Assets and are in respect of goods or services that will be provided to the Business following the Closing, including any unresolved vendor rebates or credits associated with such trade creditors that are unpaid, uncollected or unresolved, as the case may be, at the Closing;
- any Transfer Taxes applicable to the transfer of the Purchased Assets for (i) which Purchaser is responsible pursuant to Section 7.10(a);
- any Liability for any Tax or Taxes arising out of or relating to the (i) operation of the Business (as currently or formerly conducted) or the ownership of the Purchased Assets with respect to any Post-Closing Period;
- all Liabilities under the Consent Decree in connection with the Purchased Assets arising out of or relating to any act, omission, circumstance or other Event occurring after the Closing;
- all Environmental Liabilities arising out of or relating to any (i) violation, by the Purchaser or any of its Affiliates (including the Business) or their respective Representatives occurring after the Closing, of any environmental covenant or environmental deed notice recorded with respect the Purchased Assets, (ii) change in use, discontinued use, closure or shutdown of the Purchased Assets after the Closing (other than with respect to the Above-Grade Refinery Assets in connection with the dismantling, demolition or removal thereof) or (iii) the Liabilities with respect to the Above-Grade Refinery Assets assumed by Purchaser pursuant to Section 2.3(o);
- all Liabilities associated with any draw on any Seller Letter of Credit to the extent such Liabilities are exacerbated, triggered, increased or have their timing accelerated

by any act or omission of, or any delay caused by, Purchaser or any of its Affiliates, their respective successors or assigns or their respective Representatives;

- (n) Purchaser's obligations under Section 7.33; and
- (o) other than for any Excluded Refinery Liabilities, any Environmental Liabilities directly arising from the physical presence of Hazardous Materials (including asbestos and asbestos-containing materials) on, at or within (but not under, adjacent or otherwise proximate to) the Above-Grade Refinery Assets.
- 2.4 <u>Excluded Liabilities</u>. Notwithstanding anything contained herein to the contrary, Purchaser shall not assume, or cause to be assumed, or be deemed to have assumed or caused to have assumed or be liable or responsible for any of the following Liabilities of Seller or any of its Affiliates (collectively, the "**Excluded Liabilities**"):
 - (a) all Liabilities not set forth in clauses (a) (o) of Section 2.3;
- (b) all Liabilities (other than Environmental Liabilities) relating to or arising out of any breach or violation of any Law (other than any Environmental Law), Business Contract, Real Property Lease or any Business Permit occurring prior to Closing;
- (c) any Liability arising out of or relating to the Excluded Assets or any other assets of Seller or its Affiliates that are not Purchased Assets (other than any Assumed Liability);
- (d) any Liability arising out of or relating to this Agreement or any Transaction Document for which Seller has responsibility;
- (e) any severance obligations that accrue under any severance plan of Seller existing at or prior to the Closing with respect to Seller's employees and resulting from actions taken by Seller prior to the Closing;
- (f) without limiting <u>Section 2.3(h)</u>, any amounts due from Seller or its Affiliates to the USVI Government pursuant to the Concession Agreement;
- of or relating to any act, omission, circumstance or other Event occurring prior to the Closing, including (i) Seller's obligations under the RCRA Permit to the extent such obligations (A) are not exacerbated, triggered, increased, or have their timing accelerated by any act or omission of, or any delay caused by, Purchaser or any of its Affiliates, their respective successors or assigns or their respective Representatives or (B) do not arise out of or relate to a non-industrial change in use, discontinued use, long-term closure or permanent shutdown of the Purchased Assets after the Closing (other any of Seller's obligations under the RCRA Permit arising out of the dismantling, demolition or removal of the Above-Grade Refinery Assets), (ii) Liabilities arising from the MTBE Litigations (as such term is defined in the Seller Disclosure Letter), (iii) any Liabilities arising from Orders issued to Seller that may result in Environmental Liabilities to the extent such Liabilities (A) are not exacerbated, triggered, increased, or accelerated by any act or omission of, or any delay caused by, Purchaser or any of its Affiliates, their respective successors or assigns or their respective Representatives or (B) do not arise out of or relate to a change in

use, discontinued use, closure or shutdown of the Purchased Assets after the Closing (other any of Seller's obligations under such Orders arising out of the dismantling, demolition or removal of the Above-Grade Refinery Assets) or (C) are not Liabilities under the Consent Decree in connection with the Purchased Assets arising out of or relating to any act, omission, circumstance or other Event occurring after the Closing, and (iv) all Excluded Refinery Liabilities and all other Liabilities (other than Liabilities with respect to the Above-Grade Refinery Assets assumed by Purchaser pursuant to Section 2.3(0)) arising out of or relating to any act, omission, circumstance or other Event occurring prior to the Closing with respect to the Above-Grade Refinery Assets;

- any costs, expenses or other Liabilities incurred by Purchaser arising from (h) or relating to any Remedial Action (i) required to be undertaken by Seller to address an Excluded Liability pursuant to applicable Environmental Laws and (ii) not fully performed by Seller to the satisfaction of applicable Governmental Entities;
- (i) any obligations of Seller (including funding obligations) under the Pension Plan;
- any Liabilities arising pursuant to the NRD Settlement Agreement and in (i) connection with actions taken or circumstances that arose prior to the Closing Date to the extent such Liabilities (i) are not exacerbated, triggered, increased, or accelerated by any act or omission of, or any delay caused by, Purchaser or any of its Affiliates, their respective successors or assigns or their respective Representatives, or (ii) do not arise out of or relate to a change in use, discontinued use, closure or shutdown of the Purchased Assets after the Closing;
- (k) any Liabilities that both (i) arise under the Consent Decree and (ii) are in connection with actions taken or circumstances that arose prior to the Closing Date to the extent such Liabilities (A) are not exacerbated, triggered, increased, or accelerated by any act or omission of, or any delay caused by, Purchaser or any of its Affiliates, their respective successors or assigns or their respective Representatives or (B) do not arise out of or relate to a change in use, discontinued use, closure or shutdown of the Purchased Assets after the Closing;
- (1)all Liabilities for fees and expenses (i) relating to the negotiation and preparation of this Agreement and the Transaction Documents and (ii) relating to the transactions contemplated by this Agreement and the Transaction Documents, in each case, to the extent incurred by Seller or its Affiliates;
- any Liability for any Tax or Taxes of Seller or its Affiliates for any taxable (m) period;
- any Liability for any Tax or Taxes arising out of or relating to the (n) operation of the Business (as currently or formerly conducted) or the ownership of the Purchased Assets in any Pre-Closing Period, including any Property Taxes with respect to any Pre-Closing Period;
- any Liability for any withholding taxes imposed as a result of the (o) transactions contemplated by this Agreement;

- (p) the Pinnacle Cure Amount;
- (q) any Transfer Taxes applicable to the transfer of the Purchased Assets for which Seller is responsible pursuant to Section 7.10(a); and
- (r) any Liability or obligation arising out of or relating to Indebtedness of Seller or any of its Affiliates, including pursuant to the Notes.

Notwithstanding the foregoing, to the extent any of the Liabilities excluded pursuant to clauses (a) through (r) of this <u>Section 2.4</u> would cause a Liability otherwise assumed pursuant to <u>Section 2.3(n)</u> or (o) to be an Excluded Liability, <u>Section 2.3(n)</u> or (o), as applicable, shall govern and such Liability shall not be an Excluded Liability.

2.5 Assumption and Assignment of Contracts.

- (a) (i) Until five (5) Business Days prior to the Closing Date, Seller may provide updates or supplements to Section 2.1(a)(i) or Section 2.1(a)(iv) of the Seller Disclosure Letter to include revised Cure Amounts and revised amounts of any accrued and unpaid expenses and accounts payable with respect to any Business Contracts or Real Property Leases set forth therein, which updates shall amend Section 2.1(a)(i) or Section 2.1(a)(iv) of the Seller Disclosure Letter for all purposes hereof. Prior to the Sale Hearing, Seller shall commence appropriate proceedings before the Bankruptcy Court and otherwise take all reasonably necessary actions in order to determine Cure Amounts with respect to any Business Contract or Real Property Lease entered into prior to the Petition Date.
 - (ii) At the Closing, Seller shall assume and assign to Purchaser the Business Contracts and Real Property Leases, in each case pursuant to Section 365 of the Bankruptcy Code and the Sale Order, subject to provision by Purchaser of adequate assurance as may be required under Section 365 of the Bankruptcy Code and payment of the Cure Amounts in respect of Business Contracts and Real Property Leases as contemplated hereby.
 - (iii) Notwithstanding anything in this Agreement to the contrary, Purchaser may, from time to time and in its sole and absolute discretion, amend or revise Section 2.1(a)(i) and Section 2.1(a)(iv) of the Seller Disclosure Letter in order to remove any Contract from Section 2.1(a)(i) or Section 2.1(a)(iv) of the Seller Disclosure Letter, as applicable, up to five (5) Business Days prior to the Closing Date (the "Designation Deadline"); provided, that Purchaser shall not remove the Pinnacle Contract and shall, to the extent approved by the Sale Order, assume the Pinnacle Contract pursuant to Sections 2.1 and 2.3. Automatically upon the removal of any Business Contract or Real Property Lease from Section 2.1(a)(i) or Section 2.1(a)(iv) of the Seller Disclosure Letter, as applicable, by Purchaser in accordance with the first sentence of this Section 2.5(a)(iii) (any such deleted Business Contract or Real Property Lease, an "Eliminated Agreement"), it shall be an Excluded Asset for all purposes of this Agreement, and no Liabilities arising thereunder or relating thereto shall be assumed by Purchaser or be the obligation, liability or responsibility of Purchaser.

At any time and from time to time after the Closing, without further consideration, each party hereto shall, at the reasonable request of the other party hereto, execute and deliver such further instruments of conveyance, assignment, assumption and transfer with respect to the Purchased Assets and the Assumed Liabilities and take such further action as may be necessary or appropriate to (i) effectuate the intent of this Agreement, (ii) perfect or record title of Purchaser in the Purchased Assets, (iii) put Purchaser in possession of the Purchased Assets and ensure that Purchaser assumes the Assumed Liabilities and (iv) provide such other party in all material respects with the intended benefits of this Agreement. If Purchaser receives or becomes aware that it holds any of the Excluded Assets, Purchaser agrees to promptly return or cause the return to Seller of, or allow Seller or its Representatives to remove and recover, such assets at Purchaser's sole cost and expense. In furtherance of the foregoing, with respect to such Excluded Assets that Seller will remove and recover, Purchaser shall grant to Seller and its Representatives reasonable access to Purchaser's property from and after the Closing to permit Seller and its Representatives to remove and recover such Excluded Assets and make any other appropriate arrangements with respect thereto. If Seller receives or becomes aware that it holds any of the Purchased Assets, Seller agrees to promptly return or cause the return to Purchaser of, or allow Purchaser or its Representatives to remove and recover, such assets at Seller's sole cost and expense. In furtherance of the foregoing, with respect to such Purchased Assets that Purchaser will remove and recover, Seller shall grant to Purchaser and its Representatives reasonable access to Seller's property from and after the Closing to permit Purchaser and its Representatives to remove and recover such Purchased Assets and make any other appropriate arrangements with respect thereto.

2.6 <u>Required Consents.</u>

- (a) Notwithstanding anything to the contrary contained herein, to the extent that the sale, conveyance, transfer, assignment or delivery or attempted sale, conveyance, transfer, assignment or delivery to Purchaser of any Purchased Asset or Tug Boat pursuant to sections 363 and 365 of the Bankruptcy Code is prohibited by any applicable Law or Order or would require the Consent of any Third Party or any Governmental Entity and such Consent cannot be effectively overridden or canceled by the Sale Order or other related Order of the Bankruptcy Court or shall otherwise not have been obtained prior to Closing, this Agreement shall not constitute a sale, conveyance, transfer, assignment or delivery, or an attempted sale, conveyance, transfer, assignment or delivery of such Purchased Asset or Tug Boat (a "Deferred Asset") and the provisions set forth below in Section 2.6(c) shall govern; provided, however, that with respect to any Deferred Terminal Parcels, the provisions set forth in Section 7.28 shall govern. For the avoidance of doubt, nothing in this Section 2.6 shall limit or have an effect on the conditions to Closing set forth in Section 8.1(e) (in respect of the Required Governmental Approvals) or Section 8.2(d) (in respect of the Other Required Approvals).
- (b) Following the Closing, the Parties shall have a continuing obligation to use their respective commercially reasonable efforts to cooperate with each other and to obtain promptly all Consents that cannot be effectively overridden or canceled by the Sale Order or other related Order of the Bankruptcy Court and that are necessary for the transfer of the Deferred Assets and each Party shall provide all of the assistance that is reasonably requested by the other Party in connection with securing such Consents; provided, that the Parties' obligations under this Section 2.6(b) and Section 2.6(c) shall be subject to the Efforts Conditions.

To the extent that Consent is not received and cannot be effectively (c) overridden or canceled by the Sale Order or other related Order of the Bankruptcy Court in respect of any Deferred Asset as of Closing, the Parties shall reasonably cooperate with each other and use their respective commercially reasonable efforts to enter into any lawful and feasible arrangement pursuant to which Purchaser will obtain (for no additional cost or consideration) the same rights and benefits, and assume the same obligations, with respect to such Deferred Asset following the Closing, of Seller or HOVIC, as applicable, immediately prior to the Closing, whereby: (i) Purchaser shall be responsible for performing all obligations in respect of such Deferred Asset required to be performed by Seller or HOVIC, as applicable, after the Closing to the extent set forth in this Agreement; (ii) Seller or HOVIC, as applicable, shall give such commercially reasonable assistance as Purchaser may reasonably require to enable Purchaser to enforce its rights under such Deferred Asset and to ensure that Purchaser has the sole and exclusive right to direct and control the exercise or waiver of any rights under such Deferred Asset; and (iii) Seller and HOVIC, as applicable, shall promptly pay to Purchaser, when received, all monies received by Seller and HOVIC, as applicable, under any such Deferred Asset; provided, that Purchaser shall not be responsible for any Liabilities under any such Deferred Asset to the extent constituting Excluded Liabilities or arising from any negligence or misconduct by Seller, or any failure by Seller or its Affiliates to comply with Law, the terms of such Deferred Asset, this Agreement or any of the Transaction Documents. Upon obtaining the requisite Consent, Seller or HOVIC, as applicable, shall promptly convey, transfer, assign and deliver, or cause to be conveyed, transferred, assigned and delivered, such Deferred Asset to Purchaser at no additional cost.

ARTICLE III

PURCHASE PRICE

Purchase Price; Delivery of Funds. At the Closing, Purchaser shall in full consideration for the sale and transfer by Seller of the Purchased Assets and for the sale and transfer by HOVIC of the Tug Boats (a) pay to Seller an amount (such amount, the "Closing Payment") equal to (i) the Purchase Price, less (ii) the Good Faith Deposit Amount, less (iii) the Pinnacle Cure Amount, less (iv) the USVI Government Concession Fee, by wire transfer of immediately available funds to an account designated by Seller in writing to Purchaser at least three (3) Business Days prior to the Closing; (b) together with Seller, execute and deliver a joint written instruction to the Escrow Agent instructing it to release from the Escrow Account to Seller by wire transfer of immediately available funds to an account designated by Seller, an amount equal to the Good Faith Deposit Amount, and (c) assume the Assumed Liabilities.

3.2 Good Faith Escrow Deposit.

Purchaser has paid to the Escrow Agent the Good Faith Deposit Amount, which funds shall be held in the Escrow Account by the Escrow Agent and invested as provided for in the Escrow Agreement and released by the Escrow Agent only in accordance with the terms of this Agreement and the Escrow Agreement.

- (b) If the Closing occurs, then the Good Faith Deposit Amount, shall be paid to Seller and the applicable Parties shall submit joint written instructions to the Escrow Agent (in accordance with Section 3.1) to give effect to the same.
- (c) If the Closing does not occur as a result of the termination of this Agreement by Seller pursuant to Section 9.1(c), then within five (5) Business Days of such termination, the Good Faith Deposit Amount shall be disbursed to Seller by the Escrow Agent as liquidated damages and the applicable Parties shall submit joint written instructions to the Escrow Agent to give effect to the same.
- (d) If the Closing does not occur as a result of the termination of this Agreement for any reason other than as set forth in Section 3.2(c), then the Good Faith Deposit Amount shall be returned to Purchaser by the Escrow Agent within five (5) Business Days of such termination, and the applicable Parties shall submit joint written instructions to the Escrow Agent to give effect to the same.
- (e) The parties hereto acknowledge that the agreements contained in this Section 3.2 are an integral part of the transactions contemplated by this Agreement and that without these agreements neither Seller nor Purchaser would enter into this Agreement.

3.3 Allocation of the Purchase Price.

- (a) Seller and Purchaser agree to allocate the Purchase Price, Assumed Liabilities, and any other items constituting or adjusting consideration for applicable income Tax purposes (collectively, the "Tax Purchase Price") among the Purchased Assets and the Tug Boats.
- Seller shall provide Purchaser with draft allocations pursuant to Section (b) 3.3(a) that comply with Section 1060 of the Code and the Treasury regulations promulgated thereunder as soon as commercially practicable and shall use reasonable efforts to provide such draft allocations twenty (20) days prior to the Closing Date. If Seller provides Purchaser with such draft allocations twenty (20) days prior to the Closing Date then Seller and Purchaser shall use commercially reasonable efforts to reach agreement on the allocations prior to the Closing Date. If Seller does not provide Purchaser with such draft allocations at least twenty (20) days prior to the Closing Date, then Seller shall provide Purchaser with such draft allocation as soon as commercially practicable after the Closing Date (and in any event within sixty (60) days after the Closing Date). If Purchaser disagrees with the draft allocations provided by Seller, then Purchaser may, within ten (10) days after Seller provided the draft allocations, deliver written notice to Seller setting forth in reasonable detail Purchaser's objection(s) to the draft allocations. If Purchaser does not duly deliver timely written notice of any objections pursuant to the previous sentence, then Purchaser shall be deemed to have agreed to Seller's draft allocations. However, if Purchaser does duly deliver timely written notice of objections to the draft allocations, then Seller and Purchaser shall negotiate in good faith to resolve such objections and to reach agreement on the allocations; provided, that if Seller and Purchaser are not able to reach agreement within ten (10) days after Purchaser delivered its written notice of objections to Seller, then Seller and Purchaser shall submit the unresolved issues for prompt resolution to a nationally recognized accounting firm that is mutually agreed upon by Seller and Purchaser.

determination of such accounting firm shall be binding on Seller and Purchaser, and Seller and Purchaser shall each bear one-half of the costs and expenses of such accounting firm. The final allocations (as agreed upon, as deemed agreed upon or as determined by such accounting firm, as the case may be, and as updated, if applicable, pursuant to the next sentence) shall be referred to as the "Allocation." In the event of any subsequent adjustment to the Tax Purchase Price, Seller and Purchaser shall work together in good faith to agree upon an updated Allocation to reflect such adjustment, taking into account applicable Law and the nature of the circumstances giving rise to the relevant adjustment. Seller and Purchaser shall, and shall cause their Affiliates to, report consistently with the Allocation in all Returns, including IRS Form 8594, which Purchaser and Seller shall timely file with the appropriate Taxing authority, and Seller, Purchaser and each of their respective Affiliates shall not file any Return or other document or otherwise take any position that is inconsistent with the Allocation determined pursuant to this Section 3.3(b), except as may be adjusted by subsequent agreement following an audit by the Virgin Islands Bureau of Internal Revenue or the IRS or by an Order.

- (c) The Parties shall promptly inform one another of any challenge by any Governmental Entity to any allocation made pursuant to this Section 3.3 and agree to consult in good faith with and keep one another informed with respect to the state of, and any discussion, proposal or submission with respect to, such challenge; provided, that nothing in this Section 3.3 shall require Seller, Purchaser or their Affiliates to litigate before any court any proposed deficiency or adjustment by any Governmental Entity challenging the Allocation.
- Withholding. Purchaser, and any Person acting on its behalf, shall be entitled to deduct and withhold any amounts in connection with the transactions contemplated by this Agreement that are required to be deducted and withheld under applicable Law; provided, however, Purchaser shall not withhold any amounts under the Foreign Investment in Real Property Tax Act ("FIRPTA") rules if it receives the certificates described in Section 3.5(b)(viii) and in Section 3.5(d)(iii). To the extent any such amounts are deducted or withheld, such amounts shall be treated as having been paid to the Person in respect of which the deduction or withholding was made. Seller shall (and shall cause its Affiliates to) cooperate with Purchaser reasonably and in good faith, to enable Purchaser to apply for and obtain a withholding certificate on a timely basis upon which Purchaser can rely from the applicable United States and/or United States Virgin Islands Tax authorities, stating that no withholding is required in connection with the transactions contemplated by this Agreement; provided, that the Parties acknowledge and agree that (a) Purchaser's application for a withholding certificate may not result in its receipt of such certificate and Purchaser may be required to bear withholding Taxes which cannot be deducted or withheld from amounts paid to Seller under this Agreement ("Purchaser Borne Withholding Taxes") and (b) Purchaser's receipt of a withholding certificate shall not be a condition to Closing. In furtherance of and without limiting the foregoing, Seller shall promptly provide to Purchaser upon request any relevant addresses, identifying numbers, depreciation schedules, other evidence confirming adjusted basis, a calculation of the maximum tax that may be imposed in connection with the Purchase, a calculation of Seller's unsatisfied withholding liability (or evidence that no such liability exists) and any other information and documentation reasonably requested by Purchaser in connection with applying for and obtaining any such withholding certificate. In addition, Seller shall (and shall cause its Affiliates to) reasonably cooperate with Purchaser in good faith to the extent permitted by Law to apply for and obtain a refund of any Purchaser Borne Withholding Tax that

is remitted by Purchaser to a Governmental Entity in connection with the transactions contemplated by this Agreement, and to the extent Seller or its Affiliates receive a refund of any such Purchaser Borne Withholding Tax, Seller shall cause such refund to be paid to Purchaser.

3.5 Closing; Closing Deliverables.

- Subject to the satisfaction or waiver of all of the conditions set forth in (a) Sections 8.1, 8.2 and 8.3, the closing of the Purchase (the "Closing") shall take place at the offices of White & Case LLP, 1155 Avenue of the Americas, New York, New York, 10036-2787, as soon as practicable, but in any event within five (5) Business Days, after the last of the conditions set forth in Sections 8.1, 8.2 and 8.3 is satisfied or waived (other than those conditions that by their nature are to be satisfied at the Closing, but subject to the fulfillment or waiver of those conditions), or at such other time, date or place as the Parties shall agree in writing. Such date is herein referred to as the "Closing Date".
 - (b) At the Closing, Seller shall deliver or cause to be delivered to Purchaser:
 - a certificate in a form reasonably satisfactory to Purchaser signed (i) by an authorized officer of Seller, dated as of the Closing Date, confirming the matters set forth in Sections 8.2(a) and (b) with respect to Seller;
 - counterparts to the assignment, transfer and conveyance (ii) instruments listed on Section 3.5 of the Seller Disclosure Letter, in each case, duly executed by Seller (as applicable);
 - a counterpart to the Bill of Sale and Assignment and Assumption (iii) Agreement, duly executed by Seller;
 - a counterpart to the Access Easement, duly executed by Seller; (iv)
 - a counterpart to the Shared Services Agreement, duly executed by (v) Seller;
 - counterparts to the Ground Leases, duly executed by Seller; (vi)
 - a certificate, duly completed and executed by Seller and dated as of the Closing Date, in the form of Exhibit G;
 - a counterpart to each Closing Cross-Easement Agreement, duly executed by Seller;
 - a properly executed deed in recordable and customary form for conveyances of commercial real property in the United States Virgin Islands conveying the Purchased Real Property, which has the real property legal description approved by the Office of the Public Surveyor of the Government of the Virgin Islands, together with real property tax clearance letters from the Government of the Virgin Islands covering all the tax bills for all of the parcels of real property being conveyed by the deed;

Case 1:15-bk-10003-MFW Doc 394-1 Filed 12/01/15 Entered 12/01/15 12:35:34 Desc Exhibit Page 41 of 49

- (x) written instructions to the Escrow Agent to release the Good Faith Deposit Amount pursuant to Section 3.1; and
 - (xi) a counterpart to the Option Agreement, duly executed by Seller.
- (c) At the Closing, Purchaser shall deliver or cause to be delivered to Seller, and Seller shall have received:
 - (i) the Closing Payment by wire transfer of immediately available funds to an account designated by Seller in writing to Purchaser at least three (3) Business Days prior to the Closing;
 - (ii) a certificate in a form reasonably satisfactory to Seller signed by an authorized officer of Purchaser, dated as of the Closing Date, confirming the matters set forth in Sections 8.3(a) and (b);
 - (iii) counterparts to the assignment, transfer and conveyance instruments listed on <u>Section 3.5</u> of the Seller Disclosure Letter, in each case, duly executed by Purchaser (as applicable);
 - (iv) a counterpart to the Bill of Sale and Assignment and Assumption Agreement, duly executed by Purchaser;
 - (v) a counterpart to the Access Easement, duly executed by Purchaser;
 - (vi) a counterpart to the Shared Services Agreement, duly executed by Purchaser;
 - (vii) counterparts to the Ground Leases, duly executed by Purchaser;
 - (viii) a counterpart to each Closing Cross-Easement Agreement, duly executed by Purchaser;
 - (ix) written instructions to the Escrow Agent to release the Good Faith Deposit Amount pursuant to Section 3.1; and
 - (x) a counterpart to the Option Agreement, duly executed by Purchaser.
 - (d) At the Closing, HOVIC shall deliver or cause to be delivered to Purchaser:
 - (i) two (2) counterparts to the Tug Boat Bill of Sale and Acceptance with respect to each Tug Boat, in each case duly executed by HOVIC (including any notarization, apostille or other form of authentication as required in the applicable jurisdiction for such Tug Boat);
 - (ii) two (2) counterparts to the Protocol of Delivery and Acceptance with respect to each Tug Boat, in each case duly executed by HOVIC; and

- a certificate, duly completed and executed by HOVIC and dated as (iii) of the Closing Date, in the form of Exhibit G.
- At the Closing, Purchaser shall deliver or cause to be delivered to HOVIC: (e)
- two (2) counterparts to the Tug Boat Bill of Sale and Acceptance with respect to each Tug Boat, in each case duly executed by Purchaser (including any notarization, apostille or other form of authentication as required in the applicable jurisdiction for such Tug Boat); and
- two (2) counterparts to the Protocol of Delivery and Acceptance (ii) with respect to each Tug Boat, in each case duly executed by Purchaser.
- At the Closing, Purchaser shall deliver or cause to be delivered to the USVI Government, the USVI Government Concession Fee, by wire transfer of immediately available funds to an account designated by the USVI Government in writing to Purchaser.
- At the Closing, Purchaser shall deliver or cause to be delivered to Pinnacle (g) Services L.L.C., the Pinnacle Cure Amount, by wire transfer of immediately available funds to an account designated by Pinnacle Services L.L.C. in writing to Purchaser.
- The Closing shall not be deemed to have occurred unless each of the items set forth in Sections 3.5(b), 3.5(c), 3.5(d), 3.5(e), 3.5(f) and 3.5(g) has been completed or waived in writing by the applicable Party. If such items have been completed, they and the Closing shall be deemed to have been completed simultaneously. Notwithstanding anything to the contrary in this Agreement and without limiting the generality of the immediately preceding two sentences, neither Seller nor HOVIC shall acquire any right, title or interest in the funds transferred in respect of the Purchase Price, and Purchaser shall not acquire any right, title or interest in the Purchased Assets or the Tug Boats, except upon actual receipt by Seller of the Closing Payment in the bank account designated by Seller pursuant to Section 3.1.

ARTICLE IV

REPRESENTATIONS AND WARRANTIES OF SELLER

Except as set forth in the disclosure letter delivered by Seller to Purchaser (the "Seller Disclosure Letter") concurrently with the execution of this Agreement (it being agreed that any matter disclosed pursuant to any section of the Seller Disclosure Letter shall be deemed disclosed for purposes of any other section of the Seller Disclosure Letter to the extent the applicability of the disclosure to such other section is reasonably apparent on the face of such disclosure), Seller hereby represents and warrants to Purchaser as follows as of the date hereof and as of the Closing Date (or as of such other date as may be specified herein):

Due Organization, Good Standing and Limited Liability Company Power. Seller is duly organized, validly existing and in good standing under the Laws of the U.S. Virgin Islands. Seller has all requisite limited liability company power and authority to own, lease and operate its assets and properties and conduct its business as now being conducted. Seller is in good standing under the laws of each jurisdiction where the character of its assets or properties

or the conduct of its business requires such qualification, except where the failure to be in good standing would not reasonably be expected to materially and adversely affect the Business, the Purchased Assets and the Assumed Liabilities, taken as a whole, or prevent or materially delay beyond the End Date Seller's ability to consummate the transactions contemplated by this Agreement or the Transaction Documents.

4.2 Authorization; Noncontravention.

- Subject to obtaining Bankruptcy Court approval pursuant to the Sale Order, Seller has the requisite limited liability company power and authority and has taken all limited liability company action necessary to execute and deliver this Agreement, the Transaction Documents and all other instruments and agreements to be delivered by Seller as contemplated hereby, to perform its obligations hereunder and thereunder and to consummate the transactions contemplated hereby and thereby. Subject to obtaining Bankruptcy Court approval pursuant to the Sale Order, the execution, delivery and performance by Seller of this Agreement. the Transaction Documents and all other instruments and agreements to be delivered by Seller as contemplated hereby, the consummation by Seller of the transactions contemplated hereby and thereby and the performance of its obligations hereunder and thereunder have been, and in the case of documents required to be delivered at the Closing will be, duly authorized and approved by all necessary limited liability company, member or other action. This Agreement has been, and the Transaction Documents and all other instruments and agreements to be executed and delivered by Seller as contemplated hereby will be, duly executed and delivered by Seller. Subject to obtaining Bankruptcy Court approval pursuant to the Sale Order, assuming that this Agreement, the Transaction Documents and all such other instruments and agreements constitute valid and binding obligations of Purchaser and each other Person (other than Seller or any Affiliate thereof) party hereto and thereto, this Agreement, the Transaction Documents and all such other instruments and agreements constitute valid and binding obligations of Seller, enforceable against Seller in accordance with the terms thereof, except to the extent that such enforcement may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or other similar Laws affecting the enforcement of creditors' rights generally and by general equitable principles (whether considered in a proceeding in equity or at law).
- Except as a result of the Bankruptcy Case and, subject to obtaining Bankruptcy Court approval pursuant to the Sale Order, the execution and delivery by Seller of this Agreement, the Transaction Documents and all other instruments and agreements to be delivered by Seller as contemplated hereby do not, and the consummation of the transactions contemplated hereby and thereby will not:
 - conflict with any of the provisions of the certificate of formation of Seller or the Hovensa Operating Agreement;
 - except as provided in Section 4.2(b)(ii) of the Seller Disclosure Letter and subject to receipt of the Consents set forth in Section 4.3 of the Seller Disclosure Letter, conflict with or result in a breach of, or constitute a default under or give rise to any right of termination, cancellation, modification or acceleration (including any right of first refusal or similar right) or the loss of a benefit under, or require the Consent of or giving of notice to any Person under any Business Contract, Business

Permit or Real Property Lease (in each case, with or without notice or lapse or time or both); except, in the case of this clause (ii) for such conflicts or breaches or Consents or notices that (if not received) would not reasonably be expected to materially and adversely affect the Business, the Purchased Assets and the Assumed Liabilities, taken as a whole, or prevent or materially delay beyond the End Date Seller's ability to consummate the transactions contemplated by this Agreement or the Transaction Documents:

- subject to the receipt of the Consents referred to in Section 4.3 of (iii) the Seller Disclosure Letter, contravene any Law or any Order applicable to Seller or by which any of its properties or assets are bound except such contraventions that would not reasonably be expected to materially and adversely affect the Business, the Purchased Assets and the Assumed Liabilities, taken as a whole, or prevent or materially delay beyond the End Date Seller's ability to consummate the transactions contemplated by this Agreement or the Transaction Documents; or
- result in the creation or imposition of any Lien (other than a Permitted Lien) on any of the Purchased Assets.
- Governmental Consents and Approvals. Except as a result of the Bankruptcy Case and, subject to obtaining Bankruptcy Court approval pursuant to the entry of the Sale Order and the expiration, or waiver by the Bankruptcy Court, of the 14-day period set forth in rules 6004(h) and 3020(e) of the Federal Rules of Bankruptcy Procedure (the "Bankruptcy Rules"), as applicable, except as set forth in Section 4.3 of the Seller Disclosure Letter, no Consent of or filing with any Governmental Entity must be obtained or made by Seller in connection with the execution and delivery of this Agreement or any Transaction Document by Seller or the consummation by Seller of the transactions contemplated by this Agreement or any Transaction Document, except for any Consents that, if not obtained or made, would not reasonably be expected to materially adversely affect the Business, the Purchased Assets and the Assumed Liabilities, taken as a whole, or prevent or materially delay beyond the End Date Seller's ability to consummate the transactions contemplated by this Agreement or the Transaction Documents.
- 4.4 Property. (a) Other than the Real Property, Seller owns no other real property. Seller has good and valid title to the Purchased Real Property (other than any Excluded Asset), subject to Permitted Liens. There are no outstanding options, rights of first offer or rights of first refusal to purchase any Purchased Real Property or any portion thereof or interest therein. Seller has not leased or otherwise granted to any Person the right to use or occupy the Purchased Real Property or any material portion thereof, except as set forth in Section 4.4(b) of the Seller Disclosure Letter. Other than the Deferred Terminal Parcels, each parcel of Purchased Real Property constitutes a legally subdivided lot in compliance with all applicable subdivision Laws, separate from any adjoining land or improvements.
- Section 4.4(b) of the Seller Disclosure Letter contains an accurate and complete list as of the date hereof of all Real Property Leases and all other real property leases and subleases to which Seller is a party, and sets forth the role of Seller. With respect to each Real Property Lease pursuant to which Seller is a lessee or sublessee, Seller has valid leasehold interests in all leased real property described in such Real Property Lease, free and clear of any

and all Liens, except for Permitted Liens. Each Real Property Lease is a valid and binding obligation of Seller and, to the Knowledge of Seller, is enforceable against the other parties thereto in accordance with the terms thereof, in each case, except to the extent that such enforcement may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or other similar Laws affecting the enforcement of creditors' rights generally and by general equitable principles (whether considered in a proceeding in equity or at law). Except as set forth in Section 4.4(b), of the Seller Disclosure Letter, there exists no material default or event of default (with or without notice or lapse of time or both) with respect to any Real Property Lease by Seller or, to the Knowledge of Seller, by any other party thereto and Seller has not received or delivered any notice with respect to any alleged material default that has not been rescinded or completely cured. Seller has not subleased or otherwise granted any Person the right to use or occupy any material portion of the real property leased or subleased by Seller pursuant to any Real Property Lease.

- There are no assets or properties of Seller that are used or necessary for (c) the use, maintenance and operation of the Business Real Property as currently used, maintained and operated by Seller that are not included in Purchased Assets or that will not be available to Purchaser by means of the Shared Services Agreement or the Cross-Easement Agreements, except for such assets or properties that (if not so included or accessible) are not material to Seller's current use, maintenance and operation of the Business Real Property. Knowledge of Seller, there is no, and Seller has not received written notice of any, existing or threatened change in the zoning classification of any Business Real Property (or any portion thereof) from that in effect on the date of this Agreement, in each instance, which would reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect. All water, sewer, gas, electric, telephone and drainage facilities and all other utilities and public or quasi-public improvements related thereto required by Law with respect to, or required for the operation of, the Business at the Business Real Property, are installed and available to serve the Business Real Property. There are no pending or, to the Knowledge of Seller, threatened, material interruptions (except in the ordinary course) of any utility services to any material portion of the Business Real Property. The Business Real Property has sufficient access to and from dedicated streets, whether over public or private roads, for the operation of the Business, and the responsibility for maintenance of such dedicated streets has, to the Knowledge of Seller, been accepted by the appropriate Governmental Entity. No condemnation proceeding, lawsuit or administrative action or other matter affecting and adversely impairing the current use or occupancy of the Business Real Property is pending or, to the Knowledge of Seller, threatened in writing with respect to any Business Real Property which has had or would reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect. To the Knowledge of Seller, there has been no material casualty damage at any of the Business Real Property that has had or would reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect.
- Except as set forth in Section 4.4(d) of the Seller Disclosure Letter, and (d) excluding Purchased Real Property (which is governed by Section 4.4(a) above), Seller owns and has good and valid title to, or a valid leasehold interest in, all personal property included in the Purchased Assets, free and clear of all Liens, except for Permitted Liens.
 - Affiliate Transactions. Except for the Hovensa Operating Agreement and as 4.5

disclosed in Section 4.5 of the Seller Disclosure Letter and except for employment and consultant relationships and compensation, benefits, travel advances and employee or consultant loans to any officer, director, employee or consultant of Seller, in each case, in the ordinary course of business, there is no Contract or Liability relating primarily to the Business, the Purchased Assets or the Assumed Liabilities between (a) Seller, on the one hand, and (b) any Affiliate, equity holder, option holder, officer, member, partner or director of Seller, on the other hand, that remains in force and provides for obligations of any party from and after the Closing.

4.6 Material Contracts.

- Section 4.6 of the Seller Disclosure Letter sets forth (organized to refer to (a) each of the clauses in this Section 4.6) an accurate, correct and complete list of the following Contracts to which Seller is a party and which are currently in effect as of the date hereof:
 - each Contract that is not a Business Contract and that is (A) with (i) any current or former customer of, or supplier to, the Business or (B) material to the operation of Business (assuming the Business includes an Operational Terminals Business):
 - each Contract whereby Seller has created a Lien in respect of any (ii) of the Purchased Assets;
 - each Business Contract containing any covenant limiting the (iii) freedom of Seller or any of its Affiliates (A) to compete with any Person, engage in any line of business or exploit the Purchased Assets, in each case, in any geographic territory, (B) which grants to any Person any exclusivity with respect to any geographic territory, any customer or any product or service or (C) to solicit for employment, hire or employ any Person;
 - each Business Contract that requires capital expenditures or other (iv) outstanding payments to be made by Seller, in each case, following the date hereof;
 - each Business Contract involving the sharing of profits, losses, costs or liabilities with any other Person relating to the Business, the Purchased Assets or the Assumed Liabilities;
 - each Business Contract that requires (or may require in certain (vi) circumstances) in accordance with its terms the provision of credit support, collateral, a guarantee or similar financial assurance in respect of the Business, the Purchased Assets or the Assumed Liabilities;
 - each Business Contract that (A) provides services for a fixed price (vii) or maximum fee, or pursuant to any cap or similar provisions; (B) grants "most favored nation" status (or similar status) to a Person (whether in respect of pricing or otherwise) or (C) provides any performance guarantee, material rebates, discounts, incentive or volume credits;

- (viii) each Contract that involves the resolution or settlement of any actual or threatened Proceeding relating to the Business, the Purchased Assets or the Assumed Liabilities; and
- each Contract with any Governmental Entity relating to the (ix) Business, the Purchased Assets or the Assumed Liabilities.
- The Contracts required to be listed in Section 4.6(a), together with the (b) Business Contracts and the Real Property Leases are referred to as "Material Contracts." True, correct and complete copies of each Material Contract have been made available to Purchaser prior to the date hereof. Each Business Contract is a valid and binding obligation of Seller (except for any breach or default that results from the insolvency of Seller or the commencement of the Bankruptcy Case and any breach or default to be cured through the payment of the Cure Amounts) and to the Knowledge of Seller, enforceable against the other parties thereto in accordance with the terms thereof, except to the extent that such enforcement may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or other similar Laws affecting the enforcement of creditors' rights generally and by general equitable principles (whether considered in a proceeding in equity or at law). Except as set forth in Section 4.6 of the Seller Disclosure Letter and any breach or default that results from the insolvency of Seller or the commencement of the Bankruptcy Case and any breach or default to be cured through the payment of the Cure Amounts, there exists no material default or event of default (with or without notice or lapse of time or both) with respect to any Business Contract by Seller or, to the Knowledge of Seller, by any other party thereto and Seller has not received or delivered any notice with respect to any alleged material default that has not been rescinded or completely cured.
- Intellectual Property Rights and Claims. (a) Except as set forth in Section 4.7(a) of the Seller Disclosure Letter, Seller owns all right, title, and interest to or is licensed to use the Business Intellectual Property, free and clear of any Liens, other than Permitted Liens.
- Seller has not, within the last twelve (12) months made any written claim of an infringement or misappropriation by any Third Party of its rights to any Business Intellectual Property, which claim is still pending. To the Knowledge of Seller, no Third Party is infringing or misappropriating any Business Intellectual Property.
- Except for claims that have since been satisfactorily resolved or for which the statute of limitations has lapsed, Seller has not received any written notice from any Third Party challenging the right of Seller to use any of the Business Intellectual Property that would reasonably be expected to materially and adversely affect the Business, the Purchased Assets and the Assumed Liabilities, taken as a whole (assuming the Business includes an Operational Terminals Business).
- There are no pending actions, suits or arbitrations by, before or against any (d) Person or Governmental Entity for an infringement or misappropriation by Seller of any Intellectual Property owned by any Third Party which would reasonably be expected to materially and adversely affect the Business, the Purchased Assets and the Assumed Liabilities, taken as a whole (assuming the Business includes an Operational Terminals Business).

4.8 Tax Matters. Except as set forth in Section 4.8 of the Seller Disclosure Letter:

- (a) <u>Tax Returns</u>. Seller has timely filed or caused to be timely filed, taking into account any applicable extensions, with the appropriate taxing authorities all income Tax Returns and all material non-income Tax Returns with respect to the Business or the Purchased Assets. All such Returns are correct and complete in all material respects. Seller is not currently the beneficiary of any extension of time within which to file any material tax Return with respect to the Business or the Purchased Assets. No claim has ever been made by a Governmental Entity in a jurisdiction in which Seller does not file Returns that Seller is or may be subject to taxation by that jurisdiction with respect to Taxes that would be the subject of such Returns.
- (b) Payment of Taxes. All income Tax Liabilities and all material non-income Tax Liabilities of Seller due and payable with respect to the Business or the Purchased Assets, in each instance for all Pre-Closing Periods, have been timely paid, taking into account any applicable extensions, to the extent the non-payment of such Taxes could reasonably be expected to result in a Lien (other than a Permitted Tax Lien) on the Purchased Assets, adversely affect the Business, or result in Purchaser becoming liable for such Taxes. There are no pending or threatened audits, investigations, disputes, notices of deficiency, claims or other similar actions relating to Property Taxes or any other Taxes that are imposed on a periodic basis (which are not based on income) of Seller with respect to the Business or the Purchased Assets. Seller has not waived any statute of limitations in respect of non-income Taxes or agreed to any extension of time with respect to a non-income Tax assessment or deficiency.
- (c) <u>Withholding Taxes</u>. All material Taxes that Seller is (or was) required by Law to withhold or collect on or prior to the date hereof in connection with amounts paid or owing to any employee, independent contractor, creditor, equity holder or other Third Party have been duly withheld or collected, and have been timely paid over to the proper authorities to the extent due and payable, to the extent the non-payment of such Taxes could reasonably be expected to result in a Lien (other than a Permitted Tax Lien) on the Purchased Assets, adversely affect the Business, or result in Purchaser becoming liable for such Taxes.
- (d) <u>Liens</u>. There are no Liens for Taxes on any of the Purchased Assets, other than Permitted Liens.
- (e) <u>Depreciation</u>. No Purchased Asset (i) is property required to be treated as owned by another person pursuant to the provisions of Section 168(f)(8) of the Internal Revenue Code of 1954, as amended and in effect immediately prior to the enactment of the Tax Reform Act of 1986, (ii) constitutes "tax-exempt use property" within the meaning of Section 168(h) of the Code, (iii) is "tax-exempt bond financed property" within the meaning of Section 168(g) of the Code, (iv) secures any debt the interest of which is tax-exempt under Section 103(a) of the Code or (v) is subject to a "section 467 rental agreement" within the meaning of Section 467 of the Code.
- (f) The representations and warranties contained in Section 4.8(a) and Section 4.8(b) shall only apply if and to the extent Purchaser or its Affiliates could be held liable for the Taxes to which such representations and warranties relate or if the Purchased Assets could become subject to a Lien (other than a Permitted Tax Lien) in respect of such Taxes and, for the

avoidance of doubt, such representations and warranties shall not be considered inaccurate or breached to the extent the applicable representation or warranty does not apply as a result of this Section 4.8(f).

- 4.9 Employee Benefits. Set forth in Section 4.9(a) of the Seller Disclosure Letter is, as of February 15, 2015, a true and complete list of each material employee benefit plan, within the meaning of Section 3(3) of the Employee Retirement Income Security Act of 1974, as amended ("ERISA"), and each material fringe benefit, deferred compensation, severance, stock option, stock appreciation rights, incentive and bonus plan maintained or contributed to, or required to be contributed to, by Seller or its Affiliates, in each case for the benefit of Seller's employees as of such date, but excepting any such plan sponsored in whole or in part by any government, Governmental Entity or union or employee organization or any other Person (collectively, the "Employee Benefit Plans").
- Compliance with Laws. Except as set forth in Section 4.10 of the Seller Disclosure Letter, (a) Seller is not, and has not been in the past three (3) years, in material violation of any Law or Order applicable to the Business, the Purchased Assets or the Assumed Liabilities, and (b) Seller has not received any written notice alleging or, to the Knowledge of Seller, been subject to any investigation by any Governmental Entity concerning, any such material violation of a Law or Order.
- Labor Matters. Section 4.11 of the Seller Disclosure Letter sets forth each consultant of Seller as of the date hereof who is a natural person and indicates which of such consultants are former employees of Seller or any of its Affiliates (each such former employee, a "Hovensa Consultant"). Since May 1, 2015, Seller has not had any employees. Since January 1, 2015, no Hovensa Consultant has been improperly excluded from participation in any Employee Benefit Plan, and Seller has no direct or indirect liability, whether absolute or contingent, with respect to any misclassification of any Hovensa Consultant as an independent contractor rather than as an employee, except for such exclusion or misclassification that would not reasonably be expected to result in, individually or in the aggregate, a material Assumed Liability. Seller is in compliance in all material respects with the terms of the Pinnacle Contract, except for such non-compliance that would not reasonably be expected to result in, individually or in the aggregate, a material Assumed Liability.
- Environmental Matters. Except as set forth in Section 4.12 of the Seller Disclosure Letter, with respect to the Business:
- Seller is in material compliance with all applicable Environmental Laws, and Orders including the Consent Decree and NRD Settlement Agreement;
- there are no Proceedings pending, or to the Knowledge of Seller threatened in writing, against Seller that alleges any material violation of Environmental Law and Seller has not received from any Person any written notice of material violation or alleged material violation of Environmental Law;
- Seller possesses and is in material compliance with all Permits (including the RCRA Permit) required under Environmental Laws to operate the Business as currently operated and as the Business is expected to be operated following the Closing as contemplated